

BANK OF ENGLAND

Staff Working Paper No. 823 Market-implied systemic risk and shadow capital adequacy Somnath Chatterjee and Andreas A Jobst

September 2019

Staff Working Papers describe research in progress by the author(s) and are published to elicit comments and to further debate. Any views expressed are solely those of the author(s) and so cannot be taken to represent those of the Bank of England or to state Bank of England policy. This paper should therefore not be reported as representing the views of the Bank of England or members of the Monetary Policy Committee, Financial Policy Committee or Prudential Regulation Committee.



BANK OF ENGLAND

Staff Working Paper No. 823 Market-implied systemic risk and shadow capital adequacy

Somnath Chatterjee⁽¹⁾ and Andreas A Jobst⁽²⁾

Abstract

This paper presents a forward-looking approach to measure systemic solvency risk using contingent claims analysis (CCA) as a theoretical foundation for determining an institution's default risk based on the uncertainty in its asset value relative to promised debt payments over time. Default risk can be quantified as market-implied expected losses calculated from integrating equity market and balance sheet information in a structural default risk model. The expected losses of multiple banks and their non-parametric dependence structure define a multivariate distribution that generates portfolio-based estimates of the joint default risk using the aggregation technique of the Systemic CCA framework (Jobst and Gray, 2013). This market-implied valuation approach ('shadow capital adequacy') endogenises bank solvency as a probabilistic concept based on the perceived default risk (in contrast to accounting-based prudential measures of capital adequacy). The presented model adds to the literature of analytical tools for estimating market-implied systemic risk by augmenting the CCA approach with a jump diffusion process of asset changes to inform a more comprehensive and flexible assessment of common vulnerabilities to tail risks of the four largest UK commercial banks.

Key words: Systemic risk, contingent claims analysis, jump diffusion, CoVaR, systemic expected shortfall, conditional tail expectation, capital adequacy.

JEL classification: C61, C63, G01, G21, G28.

(2) Andreas (Andy) Jobst, European Department, International Monetary Fund. Email: ajobst@imf.org

This paper was completed while Andreas Jobst was Adviser at the World Bank. We would like to thank Andy Blake for very useful comments, Luis G Liceaga for graphic layout assistance, and Ken Deeley for his invaluable programming support and guidance. The views expressed in the paper are those of the authors and do not necessarily represent the views of the Bank of England, the IMF, the World Bank, their respective Boards or Management.

The Bank's working paper series can be found at www.bankofengland.co.uk/working-paper/staff-working-papers

Bank of England, Threadneedle Street, London, EC2R 8AH Email publications@bankofengland.co.uk

© Bank of England 2019 ISSN 1749-9135 (on-line)

⁽¹⁾ Somnath Chatterjee, Bank of England. Email: somnath.chatterjee@bankofengland.co.uk

1 Introduction

The global financial crisis underscored the importance of developing new and reliable approaches that can help quantify system-wide default risk of multiple institutions (beyond the traditional focus on firm-specific default risk) as an integral element in the design and implementation of macroprudential policy and surveillance. However, assessing the importance of systemic risk with a view towards enhancing the resilience of the financial sector is not a straightforward exercise and has many conceptual challenges. In particular, there is a danger of underestimating practical problems of applying systemic risk management approaches as complex modeling risks losing transparency and could fail to accommodate important differences in business models of financial institutions and regulatory approaches across countries.

This paper addresses some of these challenges in the specification of a forward-looking framework to measure systemic solvency risk based on market-implied expected losses. We use an advanced form of *contingent claims analysis* (CCA), which defines the probability of default based on the risk-adjusted balance sheet of firms whose assets are stochastic and may be above or below promised payments on its debt obligations. In the context of banks, this approach facilitates an integrated market-based capital assessment outside the accounting-based concept of regulatory capital adequacy and helps capture how the actual default risk impacts the capital assessment ("shadow capital adequacy"), which is consistent with Haldane (2011) who stated that "market-based metrics of bank solvency would be based around the market rather than book value of capital. The market prices of banks are known to offer useful supplementary information to that collected by supervisors when assessing bank health. And there is also evidence they can offer reliable advance warnings of bank distress."

We model the magnitude of the joint default risk of multiple financial banks falling into distress as a portfolio of individual market-implied expected losses (with individual risk parameters) to generate a distribution-based perspective of system-wide capital adequacy, which reflects the variability of both assets and liabilities at different levels of statistical confidence. Following the Systemic CCA (or SCCA) model (Jobst and Gray (2013)), we combine the market-implied expected losses and the dependence between them to estimate system-wide tail risk using multivariate generalised extreme value (GEV) analysis. This approach is applied to the four largest UK banks (Barclays, HSBC, Lloyds, and RBS) over a time horizon of more than 12 years covering the period of the global financial crisis (2008-2009) and the European debt crisis (2010-2011). However, the methodology can be replicated in any banking jurisdiction where bank equity trades in the secondary market. The four sample banks have different business models and account for about 85 percent of total banking sector assets. We find that systemic risk has greatly diminished in the wake of the global financial crisis, which implies a far lower probability of these banks experiencing extreme losses at the same time.

The paper is organised as follows. Section 2 provides a brief review of the literature on estimating systemic risk within banking systems. Section 3 describes the jump diffusion process and how the model parameters are used to calculate the implicit put option values of each bank. Section 4 describes the data and the empirical results relating to the determination of market-implied expected losses of individual banks. Section 5 discusses how these bank-specific expected losses can be used to generate a multivariate density of expected losses with extreme value dependence. Section 6 explains how the estimation results can inform the design of a market-based capital assessment to complement existing

prudential approaches. Section 7 concludes.

2 Related Literature

The growing literature on systemic risk measurement has responded to greater demand for models that explore the interlinkages between banks and their implications for financial stability. During the global financial crisis, interbank linkages - both directly through money markets and indirectly through derivatives and structured finance markets - became a major source of contagion. Thus, the specification of the dependence between firms helps identify common vulnerabilities to risks based on the (assumed) collective behaviour under common distress.¹

There are two strands of model development, which resonate with different policy objectives and corresponding risk measures: (i) a particular activity causes a firm to fail, whose importance to the system imposes marginal distress on other firms (or markets) ("contribution approach"), or (ii) a firm experiences losses from a significant exposure to a single (or multiple) large shock(s) to another firm, sector, country and/or currency (or demonstrates sufficient resilience to mitigate/absorb these shocks ("participation approach"). While the *contribution approach* presumes that individual failure causes system-wide distress, the *participation approach* implies greater nuance regarding the cause and effect of risk transmission. For instance, a highly leveraged institution's sudden disposal of large asset positions and investors reassessment of its default risk can significantly disrupt trading and increase asset and/or funding risk potentially causing significant losses for other firms with similar holdings. Due to differences in portfolio composition, market access and/or business factors some firms amplify the scale of impact, while others might demonstrate sufficient resilience to absorb these shocks.

Most institution-level measures of systemic risk have focussed on empirical, bivariate approaches with an implicit or explicit treatment of statistical dependence (using the historical dynamics of market prices) to determine the joint risk or expected losses based on the assumed directionality of systemic risk propagation:

(i) for the *contribution* approach - Conditional Value-at-Risk (CoVaR)² (Adrian and Brunnermeier (2016)) and related extensions, such as Component/Incremental VaR (Liao et al. (2015)), Co-Risk (Giudici and Parisi (2016)), and copula-based CoVaR (Reboredo and Ugolini (2015); Reboredo and Ugolini (2016); Karimalis and Nomikos (2018), which show the marginal contribution of firms to system-wide losses during times of individual stress, and

(ii) for the *participation* approach - Systemic Expected Shortfall (SES) (Acharya et al. (2010)) and related extensions, such as Component Expected Shortfall (CES) (Banelescu and Dumitrescu (2013)) and Systemic Capital Shortfall/SRISK (Brownlees and Engle (2011)), which show the potential losses of individual banks in response to a large drop in overall market capitalisation of the aggregate banking sector during times of stress.³, ⁴

¹Bardoscia et al. (2017) show that solvency contagion driven by the interconnectedness between financial institutions seems to have declined in the wake of the global financial crisis.

 $^{^{2}}$ See Sheu and Cheng (2012) for a detailed discussion of the CoVaR model.

³The SES and CES are complementary, with the latter being focussed on the absolute rather than the marginal "contribution" of an individual firm to system-wide losses (after accounting for differences in leverage and market capitalisation) during times of general distress in the financial system (Popescu and Turcu (2014)).

 $^{^{4}}$ Karas and Szczepaniak (2019) combine both contribution and participation approaches in their specification of a composite measure (Systemic Turbulence Measures), which includes both CoVaR and MES.

However, only a few models have incorporated a multivariate⁵ perspective using historically informed measures of association, such as Multiple Conditional VaR (MCoVaR) and Multiple Conditional Expected Shortfall (MCoES) (Bernardi et al. (2017)), and all of them define default based on a pre-specified threshold of negative shocks to market returns rather than a structural definition of default. ⁶

There are also several studies on network analysis which model how inter-linked asset holdings generate and propagate systemic risk (Allen et al. (2010)) as an alternative to distributional approaches to model complex financial relationships and project their changes during times of stress (conditional on feedback effects) (Kanno (2016)). Haldane and Nelson (2012) argue that networks can produce non-linearity and unpredictability with attendant extreme (or fat-tailed) events. Elsinger et al. (2006) and Webber and Willison (2011) analyse systemic risk in the UK banking system with a focus on interbank exposures. But both estimate banks' marginal default probabilities using the traditional diffusion process of total assets as part of the single-firm credit risk model (Merton (1974)). As will be discussed later in the paper, this approach understates market-implied expected losses and joint default risk.⁷

We propose a forward-looking, market data-based framework for estimating systemic risk to inform a more flexible solvency assessment of banks ("shadow capital adequacy") based on the price impact of investor perception (which is expressed as expected losses) rather than accounting values (which define capital adequacy). In this framework, the general default risk of the banking sector is viewed as a portfolio of individual expected losses (with individual risk parameters) where the joint probability of all banks experiencing distress simultaneously defines the magnitude of systemic risk. Expected losses are defined by observable changes in each bank's equity and equity volatility, which determine the chance of the implied asset value (including available cash flows from operations and asset sales) falling below the amount of required funding to satisfy debt payment obligations over a specified time horizon.

We specify the statistical distribution of individual expected losses at high confidence intervals in accordance with *extreme value theory* (EVT). We then estimate multivariate dependence (bankby-bank) through a closed-form specification, which accounts for the sensitivity of expected losses to common risk factors. The market-implied linkages affecting joint expected losses can deliver important insights about the joint tail risk of multiple entities during times of stress.

 $^{^{5}}$ This could be done via modelling multivariate (firm-by-firm) dependence through either a closed-from specification or the simulation of joint probabilities

⁶Novickyte and Dicpinigatiene (2018) as well as Hattori et al. (2014) provide a comprehensive review of the existing literature of measurement approaches for monitoring of systemic financial risk. Blancher et al. (2013) take stock of the current tools enlisted by IMF staff in the context of the bilateral and multilateral macroprudential surveillance and offers detailed and practical guidance on their use. The Global Financial Stability Report (International Monetary Fund (2009a) and International Monetary Fund (2009b)) provided a first overview of available analytical tools for the identification of systemic risk in the financial sector. The Office for Financial Research of the U.S. Treasury Department published a comprehensive survey of systemic risk analytics (Bisias et al. (2012)), which also organises various indicators according to a supervisory perspective and a research perspective.

⁷Di Cesare and Picco (2018) complement this conceptual distinction by including two taxonomies that help categorise systemic risk indicators used in official publications by international institutions based on their theoretical foundations and regulatory relevance. Indicators with features that are most relevant from a regulatory perpsective have (i) a high early warning capacity (ex-ante/near-coincident), (ii) are simple to implement, and (iii) can be updated at high frequency. Systemic risk indicators can also be categorised according to their underlying analytical techniques: (i) contingent claim analysis, (ii) probability and mathematical methods, (iii) interconnection analysis, and (iv) composite measures. See also Cerutti et al. (2011).

We identify changes in firm-specific and common factors determining individual default risk, their implications for the market-implied linkages between firms, and the resulting impact on the overall assessment of systemic risk based on three model properties:

(i) Drawing on the market's evaluation of a firm's risk profile. Combining market prices and balance sheet information into a measure of implied default risk links the evaluation of solvency conditions to investor perception and provides a market-based cross-validation of accounting-based measures of regulatory capital adequacy.

(ii) Controlling for common factors affecting the firm's solvency. Changes in market conditions (and their impact on the perceived risk profile of each bank) link the market-based assessment of each bank's risk profile with the risk characteristics of other banks that are subject to common changes in market conditions. Thus, measuring the time-varying dependence of expected losses connects banks implicitly to the markets in which they obtain equity funding (as the most junior, and, thus, most risk-sensitive claim on profits).⁸

(iii) Quantifying the chance of simultaneous distress. The magnitude of systemic risk is assessed based on the combined default risk of all banks during times of stress and provides a sense of relative systemic relevance of each bank. The probability of multiple firms experiencing a significant increase of expected losses simultaneously is made explicit by computing their joint probability distribution (which also accounts for the differences in the magnitude of changes in individual expected losses).

The model augments the standard option pricing formula for measuring expected losses within the multivariate set-up of the Systemic CCA framework (Jobst and Gray (2013)) with a jump diffusion component for the pricing process of the firm's asset value. This approach helps mitigate the empirical shortcomings of non-stochastic volatility in traditional single-firm credit risk models to derive more reliable estimates of expected losses. Further refinements are possible, including various simulation approaches, which might come at the expense of losing analytical tractability.⁹ In the related literature, Hanson et al. (2004) use a jump diffusion model to estimate stock market returns. Synowice (2008) uses exchange rate quotations to analyse alternative approaches for modelling asset returns using jump diffusion processes. However, there is little empirical research on incorporating jumps in the firm value process,¹⁰ much less in estimating the expected losses of financial institutions. ¹¹ This paper contributes to the literature of analytical tools for estimating market-implied systemic risk by using a jump diffusion process of changes in banks' asset values within a CCA model.

⁸The implied asset value of a bank (which underpins the estimation of implied expected losses) is modelled as being sensitive to the same markets as the implied asset value of every other bank but by varying degrees due to different capital structures (which impact the market's perception of their implied asset values).

⁹The ad hoc model by Dumas et al. (1998) can accommodate the implied volatility smile and is easy to implement but requires a large number of market option prices. The pricing models by Heston (1993) and Heston and Nandi (2000) allow for stochastic volatility, but the parameters driving these models can be difficult to estimate. Many other models have been proposed to incorporate stochastic volatility and interest rates. Other option pricing models include those based on copulas, Levy processes, neural networks, GARCH models, and non-parametric methods. The binomial tree proposed by Cox et al. (1979) spurred the development of lattices, which are discrete-time models that can be used to price any type of option - European or American, plain-vanilla or exotic.

 $^{^{10}}$ Law and Roache (2015) use the jump diffusion process to estimate the default risks for a large sample of Chinese firms.

¹¹Note that imposing a jump diffusion process for estimating the default risk poses its own set of estimation challenges. For instance, a bank's reported total liabilities change only every quarter (in the best case), which limits the number of observations for the calibration of jump risk within a given sample.

3 Estimating Market-implied Expected losses

3.1 Payoffs to equity and debt holders

A bank's asset portfolio includes loans to non-banks, interbank loans and traded securities, which are funded by debt and equity. But the actual market value of assets is not directly observable. What is observable is the market value of equity and the face value of debt for each publicly traded bank. By viewing equity as a European call option on the bank's assets with a strike price equal to the value of debt at maturity, we can make use of this information to get an estimate of the market value of assets.

As described in Gray and Malone (2008) as well as Jobst and Gray (2013), structural credit risk models view firm's liabilities as "contingent claims" on the firm's underlying assets.¹² Firms have assets (V) that change in value over time, and a fixed amount of debt (D) that is due at some point of time in the future (T). Assets of a bank are uncertain and change due to factors such as profit flows and risk exposures. Default risk over a given horizon is driven by uncertain changes in future asset values relative to promised payments on debt.

The value of the firm is split into two - that which goes to the equity holders and that which goes to the debt holders. If, at the time (T) when the debt falls due, the assets have more than enough value to repay the liabilities (D), the excess value (V - D) goes to the equity holders. If they do not, the equity holders receive nothing. This means that the equity holder has a European call option on the value of the firm's assets at maturity (T), where the payoff is either zero or the value of assets less liabilities (D) whichever is greater. The strike price is the nominal value of outstanding debt D.

If the assets are not enough to repay the liabilities, then the firm defaults, and debt holders receive the recovery value of the assets. So the value of the debt is equivalent to a risk-free debt holding plus a short put option on the value of the assets. To the extent the debt may not be repaid in full, it is deemed to be risky. Debt holders offer an implicit guarantee by absorbing losses if there is a default. Thus, they receive a put option premium in the form of a credit spread above the risk-free rate in return for holding risky debt. Their payoff at maturity, B, can be described as the nominal value of outstanding debt, D minus their put option of selling the firm's assets at a strike price D, so that

$$B = \min[V, D] = D - \max[D - V, 0].$$
(1)

Financial crises have caused equity holders and debt holders of a bank to become more uncertain about how the value of the bank's assets will evolve in the future. A bank's creditors participate in the downside risk but receive a maximum payoff equal to the face value of debt. Equity holders benefit from upside outcomes where default does not occur but have limited liability on the downside. So uncertainty in a bank's asset value has an asymmetric effect on the market value of debt and equity.

Having identified the nature of the payoffs to debt and equity holders, the next step would be to examine how the value of the bank's assets would evolve through time relative to a default barrier.

 $^{^{12}}$ Under this approach, shareholders hold a residual claim (or call option) on the firm's total asset value after outstanding liabilities have been paid off. Thus, bondholders effectively write a European put option to share holders and receive the option premium as compensation for holding risky debt; the put option premium can be observed as a credit spread above the risk-free rate. The spread increases with a higher probability of the implied asset value falling below the default barrier over the pre-defined horizon. This, in turn, depends on the volatility of the asset value, the duration of the debt claim, and the leverage of the firm.

Stochastic assets evolve relative to a distress barrier and determine the value of liabilities with implicit options. The probability that the assets will be below the distress barrier is the probability of default.

3.2 Asset evolution according to a diffusion process

In the Merton model each bank's assets V evolve according to a Geometric Brownian Motion (GBM) with *ex-ante* fixed coefficients $[\mu, \sigma]$:

$$\frac{dV}{V} = \mu dt + \sigma dW_t \tag{2}$$

where μ is the drift, which measures the average return. The stochastic component, σdW_t , models the random change in asset value in response to external effects. σ is equal to the standard deviation of the asset return. W_t is a standard Brownian motion that is characterised by independent identically distributed increments that are normally distributed with mean 0 and variance t.

In this framework, the payoff to debtholders B_t in present value terms is simply the sum of the safe claim payoff and a short position in a put option written on the firm's assets

$$B_t = De^{-rT} - \left(De^{-rT} N(-d_2) - V_t N(-d_1) \right).$$
(3)

According to the Merton (1974) model, the expected loss of the firm as given by the value of the implicit put option, is

$$P_{t_{Merton}} = De^{-rT} N(-d_2) - V_t N(-d_1), \qquad (4)$$

where $d_1 = \frac{\ln \frac{V_t}{D} + \left(\mu + \frac{\sigma^2}{2}\right)T}{\sigma\sqrt{T}}$ and $d_2 = d_1 - \sigma\sqrt{T}$.

The Merton (1974) model is effective in capturing default risk during periods of severe financial crises and particularly for banks that are on the brink of default. However, the emprical application of the Merton model for estimating the credit spreads on the risky debt of UK banks during non-crisis periods leads to poor results. This is consistent with Leland (2006), who finds that structural credit models tend to underestimate default probabilities over shorter time horizons, especially for investment grade-rated debt.¹³ Our empirical results show that using a jump diffusion model generates more realistic estimates of default risk.

3.3 Asset evolution according to a jump diffusion process

The empirical distribution of asset returns differs in many ways from the diffusion process assumed in the Black and Scholes (1973) and Merton (1974) models. Under a standard diffusion process, such as a GBM, volatility is assumed to be constant rather than stochastic. However, if the volatility (and the correlation of assets) does not change over time, the probability of a sudden large drop in asset value is very small. And if a firm cannot default unexpectedly (without being already in financial distress), the probability of default on its short-term debt will be zero as will be its credit spreads. This is not borne out empirically as the credit spreads on even investment grade short-term corporate debt will be significantly greater than zero. Jones et al. (1984) show that credit spreads on corporate bonds are much higher than what a diffusion approach would suggest. In contrast to the Black-Scholes-Merton

 $^{^{13}}$ During the height of the crisis, several banks witnessed sharp declines in the market value of their assets over short time intervals, which are not captured by a standard diffusion process.

(BSM) framework, a reduced-form approach would treat default as a random Poisson event involving a sudden loss in market value that cannot be predicted. Despite the wide application of continuous stochastic volatility models, it has become increasingly evident that a pure diffusion process is not suitable for most asset returns. Instead, reasonable assumptions need to be made about the dynamics of the underlying pricing process, such as the incorporation of jumps.

In our paper, we add a jump process to the underlying diffusion process of the BSM framework. Jump diffusions can incorporate rare, large fluctuations in asset prices as witnessed during the global financial crisis. Moreover, jumps allow higher moment features such as skewness and leptokurtic behaviour in the distribution of asset price returns. Various specifications of jump diffusion models have been introduced to empirical finance. A short review can be found in Synowiec (2008). We apply the Merton (1976) jump diffusion model with time invariant coefficients, constant volatility and lognormally distributed jump sizes. Changes in asset prices consist of a continuous diffusion component which is modelled as a GBM, and a discontinuous jump component, which is modelled as a Poisson process. The continuous time stochastic process for the asset value V_t is given by the stochastic differential equation,

$$\frac{dV_t}{V_t} = \mu dt + \sigma dW_t + d(\sum_{q=0}^{N_t} (J_q - 1))$$
(5)

where the last term models the jumps. The jumps capture the price impact of extreme events, which arrive only at discrete points in time and these arrivals are described by a Poisson process N_t , characterised by its intensity λ . A jump is modelled by a random variable J, which transforms the asset value V_t to JV_t . The difference (J-1) is the relative change in price when a Poisson jump occurs. The jump size J_q is a sequence of independent identically distributed nonnegative random variables. In the absence of outside news, the asset price simply follows a GBM. In the model all sources of randomness, N_t , W_t and J's, are assumed to be independent. Solving the stochastic differential equation (5) gives the dynamics of the asset price:

$$V_t = V_0 \exp\{(\mu - \frac{1}{2}\sigma^2)t + \sigma W_t\} \prod_{q=0}^{N_t} J_q$$
(6)

where V_0 is the asset price at time zero. If we denote $Y_q = \log J_q$, we have

$$X_t = \log \frac{V_t}{V_o} = (\mu - \frac{1}{2}\sigma^2)t + \sigma W_t + \sum_{j=0}^{N_t} Y_q,$$
(7)

The jump size is drawn randomly from a distribution with probability density function J(Q) which is independent of both the GBM and the Poisson process. The jump size J is a log-normally distributed random variable:

$$\ln J \sim N(\mu_q, \sigma_q^2). \tag{8}$$

The expected value of the jump size can be written as:

$$E[J-1] = \exp(\mu_q + \frac{\sigma^{2_q}}{2}) - 1.$$
(9)

Augmenting the diffusion process with jumps adds three extra parameters $(\lambda, \mu_q, \sigma_q)$ to the BSM framework which contains two parameters (μ, σ) . Using the approach of Ball and Torous (1983) and discretising over $[t, t + \Delta]$, the solution takes the form

$$\Delta X_t = (\mu - \frac{1}{2}\sigma^2)\Delta + \sigma\Delta W_t + \sum_{j=0}^{\Delta N_t} Y_q,$$
(10)

where $\Delta W_t = W_{t+\Delta} - W_t \sim N(0, \Delta)$, where $\Delta N_t = N_{t+\Delta} - N_t$ is the number of jumps occuring in the interval $[t, t+\Delta]$.

The estimated asset values for each bank are transformed into log returns, $\Delta [ln(V_t)] = ln(V_{t+1}) - ln(V_t)$. For estimation purposes, we need the probability density function of X_t as in (7). Since the calibration is done in discrete time we work with ΔX_t as defined in (10) so that the density function now has a finite number of terms. The approximation assumes that $\lambda \Delta$ converges to zero; this type of discrete time specification is referred to as a Bernoulli diffusion model. When the time difference Δt is small, by the properties of the Poisson process N_t , we know that $P(\Delta N_t = 0) = 1 - \lambda \Delta t$, $P(\Delta N_t = 1) = \lambda \Delta t$ and $P(\Delta N_t > 1) = 0$. Therefore, the density in log returns, $f_{\Delta X_t}$ can be thought of as a Bernoulli random variable that has a mixture distribution for a small Δt . During Δt , the density in log returns is a weighted average of the diffusion density $(f_{\Delta Diff})$ and jump density $(f_{\Delta Jump})$ given by:

$$f_{\Delta X_t} = (1 - \lambda \Delta t) f_{\Delta Diff} + \lambda \Delta t (f_{\Delta Diff} * f_{\Delta Jump}).$$
⁽¹¹⁾

In equation (11) the diffusion part of the process is:

$$f_{\Delta Diff} \sim N((\mu - \frac{\sigma^2}{2})\Delta t, \sigma^2 \Delta t),$$
 (12)

and the jump part is:

$$(f_{\Delta Diff} * f_{\Delta Jump}) \sim N([(\mu - \frac{\sigma^2}{2})\Delta t + \mu_q], \sigma^2 \Delta t + \sigma_q^2)$$
(13)

Denoting the distribution of daily log returns in a bank's asset values as $\Delta X = \Delta \ln V_t$, the loglikelihood function over the set of parameter values $\theta = \{\mu, \sigma, \lambda, \mu_q, \sigma_q\}$, can be written as:

$$\log L(\theta \mid \Delta x_1, \dots \Delta x_T) = \sum_{t=1}^T \log f_{\Delta x}(\Delta x_t \mid \theta)$$
(14)

3.4 Estimating the parameters of the jump diffusion process

The normal practice for estimating the parameters would be to maximise the log-likelihood function over a set of given parameter values $\{\mu, \sigma, \lambda, \mu_q, \sigma_q\}$. However, in this case, the standard maximum likelihood (ML) procedure is not valid as the log return in the Bernoulli diffusion model is a mixture of two normal distributions with different means and variances as given by equations (12) and (13). The likelihood function is not well-behaved, which indicates the occurrence of discontinuous jumps. Moreover, since the intensity parameter, λ , is unknown *ex ante*, it is not possible to identify from which of the two normal distributions each observation originates. This, coupled with the fact that the two normal distributions are different, the ML estimator does not exist.¹⁴ Maximising the log-likelihood function, according to equation (14) fails to determine robust parameter estimates since the likelihood equation has a flat surface. A solution to the problem would be to devise a method that would not require all the five parameters to be optimised simultaneously.

We adopt a two-stage estimation process to calibrate the model parameters. The most intuitive way to calibrate the model would be to use a bank's values. However, these are not directly observable. If we assume that the bank's underlying market value exhibits characteristics similar to those of the observable market capitalisation, we can estimate the jump parameters $\{\lambda, \mu_q, \sigma_q\}$ directly from the available time series data. The remaining diffusion parameters $\{\mu, \sigma\}$ are then estimated using ML. This two-stage estimation process is described in the following two sections.

3.5 Detecting jumps as change points in market capitalisation returns

We estimate the jump parameters $\{\lambda, \mu_q, \sigma_q\}$ using the observable market capitalisation, of each sample bank, over an estimation window of six months from 2005 to 2017. We expect structural breaks of the time series when the mean and/or standard deviation of the series undergo significant changes. Figure 1 shows the daily log returns in the market capitalisation of a major UK bank in 2008H2 (July to December). We can see that the first break in the mean log return occurs around day number 57 (18 September 2008). However, for our estimation, we need to identify the precise time when the mean series changes abruptly, which we call "change points." A change point reflects a discontinuity or jump in the time series of log returns.

3.5.1 Model specification

Let x_i denote the log returns in the market capitalisation of a bank at time t where $1 \le i \le n$ and n = 130 days. We assume that these log returns fluctuate around some underlying signal m that could be associated with the factors driving the market capitalisation of the bank. x_i can be expressed as

$$x_i = m(t_i) + e_i,\tag{15}$$

where e_i is a sequence of residual errors of the signal against the modelled changes. Figure 1 shows that the mean does not change during some time periods, which implies that m is piecewise constant within any of these "distinct states." The horizontal lines in Figure 2 capture those distinct states whose starting and end dates are identified by a distinct change point. The objective of this detection method is to determine these change points. For the log-return time series data $x_1, ..., x_n$ if a change point occurs at τ , then $x_{1,...,x_{\tau}}$ will differ from $x_{\tau+1,...,x_n}$ in some way. Following Lavielle (2005) and Killick et al. (2012), we assume that log-returns x_i follow a normal distribution, where the means m_i are piecewise constant through time. Moreover, we assume that there exists discontinuity instants $\mu_1, \mu_2, ... \mu_K$ such that

$$m(t) = \mu_k \text{ if } \tau_{k-1} < i \le \tau_k, \tag{16}$$

¹⁴For further clarification see Kiefer (1978), Honore (1998), and Hamilton (1994).

Figure 1: UK Banking Sector (Bank 1): Daily Log Returns of Market Capitalisation, July-December 2008



Figure 1. UK Banking Sector (Bank1): Daily Log Returns of Market Capitalisation, July-December 2008

Sources: Bank of England, Bloomberg L.P., and authors.

where k - 1 is the number of change points, which gives k homogeneous intervals where the mean of the log-returns are constant, and where $\tau_0 = 0$ and $\tau_k = n$. Thus, for any $\tau_{k-1} < i \leq \tau_k$,

$$x_i = \mu_k + e_i. \tag{17}$$

The sequence of residual errors $e_i, 1 \le i \le n$ is a sequence of random variables with zero mean. So x_i is a sequence of random variables with piecewise constant mean

$$E(x_i) = \mu_k \text{ if } \tau_{k-1} < i \le \tau_k. \tag{18}$$

Assuming that the sequence of residual errors is a sequence of independent and identically distributed Gaussian variables $e_i \sim N(0, \sigma^2)$. It follows that

$$x_i \sim N(0, \sigma^2) \text{ if } \tau_{k-1} < i \le \tau_k.$$

$$\tag{19}$$

Therefore, identifying the number and sequence of jumps in the time series of log asset value returns would involve estimating (i) the number of K segments, (ii) the location of the discontinuities $(\tau_k, \text{ where } 1 \leq k \leq K - 1)$ and (iii) the value of the underlying signal or mean in each segment $(\mu_k, 1 \leq k \leq K)$. We derive these estimates by minimising the residual errors, which is equivalent to maximising the likelihood. Adding change points decreases the residual error but can result in overfitting. In the extreme case, every point becomes a change point, and the residual error vanishes.

The model is specified as a parametric model which depends on a vector of parameters $\theta = (\mu_{1,...,\mu_{K}}, \sigma^{2}, \tau_{1}, ..., \tau_{K-1})$. Since the data from each segment represent an independent set of random variables, the overall likelihood function is a product of local likelihood functions. The overall likelihood function is denoted by $L(\theta \mid x_{1}, x_{2}, ..., x_{n})$ and is obtained by multiplying all the local probability distributions $p(x_{1}, x_{2}, ..., x_{n}; \theta)$ so that

$$L(\theta \mid x_1, x_2, ..., x_n) = p(x_1, x_2, ..., x_n; \theta)$$

$$= \prod_{k=1}^{K} p(x_{\tau_{k-1}+1,\dots,x_{\tau_{k}}} : \mu_{k}, \sigma^{2})$$
$$= \prod_{k=1}^{K} (2\pi\sigma^{2})^{\frac{-(\tau_{k}-\tau_{k-1})}{2}} \exp\left\{-\frac{1}{2\sigma^{2}} \sum_{i=\tau_{k-1}+1}^{\tau_{k}} (x_{i}-\mu_{k})^{2}\right\}$$
(20)

The ML estimation of θ can be decomposed into two steps:

(i) the means (μ_k) and change points (τ_k) are estimated by minimising

$$J(\mu_1, ..., \mu_K, \tau_1, ..., \tau_{K-1}) = \sum_{k=1}^K \sum_{i=\tau_{k-1}+1}^{\tau_k} (x_i - \mu_k)^2,$$
(21)

(ii) the variance σ^2 is estimated as the empirical variance of the estimated residuals

$$\hat{\sigma}^2 = \frac{1}{n} \sum_{k=1}^{K} \sum_{j=\hat{\tau}_{K-1}+1}^{\hat{\tau}_k} (x_i - \hat{\mu}_k)^2.$$
(22)

We will focus on the critical first step, i.e., the minimisation of $J(\mu_1, ..., \mu_K, \tau_1, ..., \tau_{K-1})$. For a given sequence of change points $\tau_1, ..., \tau_{K-1}$, J can be minimised with respect to 1; ...; K. This is described below:

$$\hat{\mu}_k\left(\tau_{k-1},\tau_k\right) = \overline{x}_{\tau_{k-1}+1:\hat{\tau}_k}$$

$$= \frac{1}{\tau_k - \tau_{k-1}} \sum_{i=\tau_{k-1}=1}^{\tau_k} x_i$$

minimises $\sum_{j=\tau_{k-1}+1}^{\tau_k} (y_j - \mu_k)^2$. We insert the estimated mean values $(\mu_k (\tau_{k-1}, \tau_k))$ into J so that

$$U(\tau_1, ..., \tau_{K-1}) = J(\hat{\mu}_1(\tau_0, \tau_1), ..., \hat{\mu}_K(\tau_{K-1}, \tau_K), \tau_1, ..., \tau_{K-1})$$

$$= \sum_{k=1}^{K} \sum_{i=\tau_{k-1}+1}^{\tau_k} \left(x_i - x_{\tau_{k-1}+1:\tau_k} \right)^2.$$

The ML estimation process involves minimises

$$U(\tau_1, ..., \tau_{K-1}) = \sum_{k=1}^{K} \sum_{j=\tau_{k-1}+1}^{\tau_k} \left(x_i - \overline{x}_{\tau_{k-1+1}:\tau_k} \right)^2.$$
(23)

We use a dynamic programming algorithm for solving this optimisation problem, ¹⁵ which is explained in Lavielle (2017). In our analysis, we select the optimum number of change points k (where k = 1, 2, 3, ..., 10) to minimise the residual returned by the model. Specifying a maximum number of k change points does not guarantee that k change points will be found. Rather, any number of change points from 1 up to k could be found.

 $^{^{15}}$ We have implemented the algorithm in MATLAB programming language by applying the *findchangepts()* function.

Figure 2: UK Banking Sector (Bank 1): Daily Log Returns of Market Capitalisation with Identified Change Points, July-December 2008



Figure 2. UK Banking Sector (Bank 1): Daily Log Returns of Market Capitalisation with Identified Change Points, July-December 2008

Sources: Bank of England, Bloomberg L.P., and authors. Note: number of change points = 7, total residual error = 0.31935.

As shown in Figure 2, the output from the algorithm tells us the number of times the mean of the log returns series x_i changes most significantly and also the dates at which those changes occur. In this case there were 7 such identified change points, which indicate the number of jumps in the time series of the log returns for the period under observation. The algorithm also returns the residual error of the signal against the modelled changes which, in this case, is 0.31935. Figure 2 shows the results from the algorithm for a UK bank between July and December 2008.

3.5.2 Jump diffusion parameters

Having identified the number of jumps within the time series of log returns in market capitalisation, we determine the jump rate λ , as the number of jumps divided by the total number of observations. For sample bank (Bank 1) shown in Figure 2 this was 0.0534 (7 divided by 131). The corresponding mean jump size $\hat{\mu}_q$ is -0.0456, and the jump size volatility $\hat{\sigma}_q$ is 0.2329. As expected, during the financial crisis period July-December 2008, the mean jump size of the daily log returns in market capitalisation for all the four sample banks turned out to be negative.

3.6 Calibrating the remaining parameters of the jump diffusion model

Having obtained estimates of the jump parameters $\{\lambda, \mu_q, \sigma_q\}$, we now estimate the remaining parameters μ and σ using conditional ML. We re-specify the negative of the log-likelihood for the mixed density function in equation (11) such that the parameter values, θ^* , can be written as:

$$\theta^* = \left\{ \mu, \sigma, \hat{\lambda}, \hat{\mu}_q, \hat{\sigma}_q \right\}.$$
(24)

This indicates that the jump parameters are now held constant within the log-likelihood function for a bank's returns, and what varies is σ and μ . However, the value of the likelihood function is ultimately determined by a single unknown parameter, σ . According to the standard Merton (1974) model, the diffusion mean

$$\mu = \frac{\left(\log V_{t+\triangle} - \log V_t\right)}{T} + \frac{1}{2}\sigma^2 \tag{25}$$

is dependent on the diffusion volatility.

In order to calibrate the jump diffusion model, we choose exogenously an initial value of the unobserved volatility of asset value returns σ . A bank's actual asset value V is not observable. However, if a bank's shares are publicly traded, we can observe its market value, which is reflected in the price of equity E. So the estimation process begins by fitting the jump diffusion model to the observed time series of bank equity prices (market capitalisation series) and then producing an initial estimate for the market value over the residual maturity T - t so that

$$E_t = C_t = \sum_{k=0}^{\infty} \frac{e^{-\lambda T} (\lambda T)^k}{k!} [V_t e^{\mu T + k(\mu_q + \frac{\sigma_q^2}{2})} N(d_1) - D e^{-rT} . N(d_2)],$$
(26)

where

$$d_{1} = \frac{\ln \frac{V_{t}}{D} + (\mu + \frac{\sigma^{2}}{2})T + k(\mu_{q} + \sigma^{2}_{q})}{\sqrt{\sigma^{2}T + k\sigma^{2}_{q}}}$$
(27)

$$d_2 = d_1 - \sqrt{\sigma^2 T + k\sigma_q^2}.$$
(28)

Expected loss is modelled as a put option where the present value of debt D represents the strike price

$$P_t = \sum_{k=0}^{\infty} \frac{e^{-\lambda T} (\lambda T)^k}{k!} [De^{-rT} . N(-d_2) - V_t e^{\mu T + k(\mu_q + \frac{\sigma_q 2}{2})} N(-d_1)].$$
(29)

This assumes a more realistic evolution process of a bank's asset value than the specification of expected losses according to the traditional Merton model in equation (4). The market value of the bank (V) comprises the equity (E) and the market value of debt (B) at time t. It can be represented as:

$$V_t = E_t + B_t. \tag{30}$$

Put-call parity states that owning the asset V_t outright is equivalent to owning a portfolio comprising (i) a call option C_t at strike price D, (ii) a risk-free bond valued D at time T, and (iii) a short put option P_t with strike price D so that

$$V_t = C_t + De^{-rT} - P_t. (31)$$

Rearranging we have:

$$C_t = V_t - (De^{-rT} - P_t). (32)$$

In equation (32), $De^{-rT} - P_t$ is the market value of debt B_t . As an initial step, we solve for V_t using the jump diffusion parameters estimated from the observable changes in equity and equity volatility. After updating the parameter estimates based on this solution for V_t , we follow an iterative procedure until the estimates of V_t , μ , σ , λ , μ_j and σ_j have converged. After convergence, we can plot the negative log-likelihood surface in a neighbourhood of the candidate solution point to verify that a local minimum point was identified by the ML process. Figure 3 plots such a surface for the same UK bank (shown in Figure 2) in 2008H2. The charts in Appendix A show the estimated parameter values of the diffusion volatility σ , for each of the four sample banks together with their conditional confidence intervals. For the parameter σ , the 95% confidence interval is defined as:

$$\left[\hat{\sigma} - 1.96 \times SE\left(\hat{\sigma}\right), \hat{\sigma} + 1.96 \times SE\left(\hat{\sigma}\right)\right].$$
(33)

Following Zhou (1997), we can calculate the risk-neutral default probability for each individual bank by denoting the drift μ_j by r.

$$\Pr\{\frac{V}{D} \le \varsigma\} = \sum_{k=0}^{\infty} \frac{e^{-\lambda T} (\lambda T)^k}{k!} \cdot N(\frac{\ln(\varsigma) - \ln(\frac{V}{D}) - (r - \frac{\sigma^2}{2} - \lambda v)T - k\mu_q}{\sqrt{\sigma^2 T + k\sigma_q^2}}),$$
(34)

where ς is the default barrier, which is set equal to one. The time to maturity of the nominal debt

Figure 3: UK Banking Sector (Bank 1): Log-Likelihood Curve, July-December 2008





Sources: Bank of England, Bloomberg L.P., and authors.

outstanding is assumed to be one year. We define D to be the sum of a bank's total short-term liabilities and half of long-term liabilities to reflect the fact that not all of the longer maturity debt will need to be repaid within the course of a year (Crosby and Bohn (2003)). The expected loss can be decomposed into the probability of default (PD) and the loss given default (LGD),

$$P_{t} = \underbrace{\sum_{k=0}^{\infty} \frac{e^{-\lambda T} (\lambda T)^{k}}{k!} N(-d_{2}) \left[1 - \frac{N(-d_{1})}{N(-d_{2})} \frac{V_{t} e^{\mu T + k(\mu_{q} + \frac{\sigma_{q}^{2}}{2})}}{De^{-rT}} \right]}_{PD} \underbrace{De^{-rT}}_{LGD},$$
(35)

and under the Merton (1974) model

$$P_{tMerton} = \underbrace{N(-d_2) \left[1 - \frac{N(-d_1)}{N(-d_2)} \frac{V_t}{De^{-rT}} \right]}_{PD} \underbrace{De^{-rT}}_{LGD}.$$
(36)

Equation (35) shows that the LGD is endogenously determined in this model. Banks default when they are not able to pay their debt in full. In turn, creditors that fail to receive some payments from their counterparties might not be able to pay their own creditors, potentially triggering a cascade of defaults. Creditors do not necessarily lose the full face value of their claims towards defaulted banks if the recovery rate is positive.

4 Data and Results

The model is calibrated using market information and balance sheet data of four major UK banks. Data about the amount of outstanding debt and the corresponding maturity structure are sourced from published semi-annual accounts (2005 H1 to 2017 H2), consistent with the statutory reporting cycle. We obtain daily observations on equity prices and the secondary market yields on one-year UK Treasury bonds as the risk-free interest rate. The combined panel dataset comprises observations for each bank j over a six-month period. Based on the time series of observed equity prices and the risk-free rate, the jump diffusion process given by equation (10) is used to back out a corresponding series of implied asset values over a 13-year study period from 3 January 2005 to 30 December 2017 (with a total number of 3,390 observations) for each bank over six-month rolling windows. The expected losse of a bank is the equivalent of the put option premium specified in equation (35). The expected losses of all banks peaked in 2008 and remained elevated until the end of 2009. There was another spike following the European debt crisis during the second half of 2011. Figure 4 plots the expected losses (in billions of Pound Sterling) of the four sample banks from 2005 to 2017 estimated based on the specification in equation (35) above.

Figure 4: UK Banking Sector (Largest Listed Banks): Individual Expected Losses (Put Option Values with Jump Diffusion), 2005-17 (GBP Billions)





Sources: Bank of England, Bloomberg L.P., and authors.

Figure 5: UK Banking Sector (Largest Listed Banks): Individual Expected Losses (Put Option Values without Jump Diffusion) 2005-17 (In GBP Billions)



Figure 5. UK Banking Sector (Largest Listed Banks): Individual Expected Losses (Put Option Values without Jump Diffusion), 2005-17 (In GBP Billions)

Sources: Bank of England, Bloomberg L.P., and authors.

Figure 6: UK Banking Sector (Largest Listed Banks): Comparison of Expected Losses for Bank1 (with and without a Jump Diffusion of asset values) 2005-2017 (In GBP Billions)





Sources: Bank of England, Bloomberg L.P., and authors.

Figure 5 shows the estimation results for expected losses consistent with the Merton (1974) model as specified in equation (36). While this simpler option pricing approach still captures the surge of expected losses of banks during both the global financial crisis and the European soevereign debt crisis, there is a distinct difference in the order of magnitude and timing. More importantly, sporadic stresses are not picked up under this conventional estimation of expected losses, suggesting very low credit spreads on outstanding debt outside distress periods. Figure 6 makes the comparison for the expected losses of a single bank (Bank 1), which confirms the different estimation results obtained with and without a jump diffusion of asset values underpinning the option pricing approach.

5 Estimating Joint Expected Losses

In this section, the individually estimated expected losses obtained from the previous analysis are combined to determine the magnitude of default risk on a system-wide level using their dependence structure. As part of the valuation of the different risks affecting a bank's operating performance, expected losses can be modeled by explicitly taking into account extreme events. Based on the empirical evidence presented in the previous section, the distribution of expected losses is assumed to be "fat-tailed" in keeping with *extreme value theory* (EVT). For instance, closed-form methods are frequently used to help define the limiting behaviour of extreme observations. ¹⁶Estimating joint expected losses of multiple institutions, however, entails a non-trivial aggregation problem. Since the simple summation of individual expected losses would presuppose perfect correlation, i.e. a coincidence of defaults, the correct estimation of aggregate risk requires knowledge about the dependence structure of individual balance sheets and associated expected losses. While it is necessary to move beyond "singular CCA" by accounting for the dependence structure of individual balance sheets, the estimation of systemic risk through correlation becomes exceedingly unreliable in the presence of "fat tails".

Correlation describes the complete dependence structure between two variables correctly only if the joint distribution is elliptical - an ideal assumption rarely encountered in practice. This is especially true in times of stress, when default risk is highly skewed, and higher volatility inflates conventional correlation measures automatically (as covariance increases disproportionately to the standard deviation). Alternative measures of dependence between risk factors can capture nonlinear dynamics of changes in variables far removed from the median.¹⁷ Copula functions and other nonparametric methods provide the possibility to combine two or more distributions of variables based on a more flexible specification of their dependence structure at different levels of statistical significance. These approaches generate measures of so-called "joint asymptotic tail dependence," which define the expectation of common extreme outcomes.¹⁸

As part of a four-step process, we define the univariate marginal density of all firms, which are then combined with their dependence function in order to generate an aggregate measure of default risk as

¹⁶The generalised extreme value (GEV) distribution specifies the asymptotic tail behaviour of the order statistics of normalised maxima (or minima) drawn from a sample of dependent random variables, whereas the generalised Pareto distribution (GPD) is an exceedence function, which measures the residual risk of extremes beyond a given threshold (that is, a designated maximum (or minimum) as the conditional distribution of mean excess).

¹⁷For example, an expedient nonparametric method of investigating the bivariate empirical relation between two random vectors is to ascertain the incidence of shared cases of cross-classified extremes via a quantile-based statistic of independence.

¹⁸See also Gray and Jobst (2010) and Gray and Jobst (2011).

joint expected losses based on the the multivariate set-up underpinning the Systemic CCA framework.¹⁹ The dependence function combines several marginal distribution functions in accordance with Sklar's theorem (Sklar (1959)) on constructing joint distributions with arbitrary marginal distributions via copula functions (which completely describes the dependence structure and contains all the information to link the marginal distributions).²⁰ We can then use tail-risk estimates such as *conditional tail expectation* (CTE) (which is sometimes referred to as *conditional Value-at-Risk (VaR)* or *expected shortfall*) in order to gauge systemic solvency risk in terms of stress at a statistical confidence level of choice.

5.1 Estimating the marginal distributions of individual expected losses

We first specify the statistical distribution of individual expected losses (based on the time series of put option values obtained for each bank) in accordance with EVT, which is concerned with modeling the tails of probability distributions where extreme risks are situated. Let the vector-valued series

$$X_{i}^{n} = P_{1}^{n}(t), ..., P_{m}^{n}(t)$$
(37)

denote a sequence of independent and identically distributed (i.i.d) random observations of expected losses (i.e., a total of *n*-number of daily put option values $P_m^n(t)$ up to time *t*), each estimated according to equation(29) above over a rolling window of $\tau = 120$ observations (covering half a year of working days) with period updating for $j\epsilon m$ banks in the sample over a 13-year study period from 3 January 2005 to 30 December 2017 (with a total number of 3,390 observations per bank). We define the cumulative distribution F of expected losses of *m*-banks as an ordered *n*-sequence of sample maxima based on

$$X = \max(P_1^n(t)..., P_m^n(t)).$$
(38)

Since the actual distribution is unknown, we focus on the asymptotic distribution of X, which is modelled in accordance with Fisher and Tippett (1928), which defines the attribution of a given distribution of normalised maxima (or minima) to be of extremal type (assuming that the underlying function is continuous on a closed interval). In order to obtain a non-degenerate limit for the maxima, we standardise the random variable X, by means of an affine transformation with a location parameter α_j^n and a positive scale parameter $\beta_j^n > 0$, thereby focussing on the asymptotic distribution of $(X - \alpha^n)/\beta^n$.

The probability of each ordered *n*-sequence of normalised sample maxima $(X - \alpha^n)/\beta^n$ converges to the non-degenerate limit distribution H(x) as $n \to \infty$ and $x \in \mathbb{R}$, so that

$$F^{[\beta^n + \alpha^n]}(x) = \lim_{n \to \infty} \Pr((X - \alpha^n) / \beta^n \le y) = [F(\beta^n y + \alpha^n)]^n \to H(x)$$
(39)

 $^{^{19}}$ Jobst and Gray (2013) explain how the Systemic CCA framework can also be applied in stress testing. Any capital shortfall would arise when the level of regulatory capital after accounting for joint expected losses causing capital levels to fall below the hurdle rate of minimum capital requirement.

 $^{^{20}}$ Note that the analysis of dependence in this approach is completely different from the analysis of marginal distributions, and, thus differs from the classical approach, where multivariate analysis is performed jointly for marginal distributions and their dependence structure by considering the complete variance-covariance matrix, such as the MGARCH approach.

falls within the maximum domain of attraction (MDA) of the generalised extreme value (GEV) distribution, which defines the limiting distribution of maxima of dependent random variables (Coles et al. (1999); Poon et al. (2003); Jobst (2007)). Assuming the existence of a whole sequence (α_j^n, β_j^n) of such parameters indexed by n, Gnedenko (1943) proved that the distribution of the sequence of standardised maxima converges to one of the following three non-degenerate types of distributions:

$$H_0(x) = \exp\{-\exp(-x)\}, x \in \mathbb{R}$$

$$\tag{40}$$

$$H_{1,\xi}(x) = \exp(-x^{-1/\xi}) \text{ if } x \in (\mu - \sigma/\xi, \infty), \xi > 0, \text{ and}$$
 (41)

$$H_{2,\xi}(x) = \exp(-(-x)^{-1/\xi}) \text{ if } x \in (-\infty, \mu - \sigma/\xi], \xi < 0.$$
(42)

The parameter μ corresponds to the scale, and σ is a location parameter; the third parameter ξ is called the shape parameter or tail index and accounts for the behaviour of the tail of the distribution. The larger the tail index, the thicker the tail, i.e., the larger the weight of the tail and the slower the speed at which the tail approaches this limit. The shape parameter also indicates the number of moments of the distribution, e.g., if $\xi = 1/2$, the first moment (mean) and the second moment (variance) exist, but higher moments have an infinite value.²¹ The upper tails of most (conventional) limit distributions, i.e., the ordered maxima, always (weakly converge) to this parametric specification of asymptotic behaviour, irrespective of the original distribution of observed maxima (unlike parametric VaR models).

A standard form of the cumulative GEV distribution function combines the three sub-models to

$$H_{\mu,\sigma,\xi}(x) = \left\{ \begin{array}{l} \exp\left(-\left(1+\xi\frac{(x-\mu)}{\sigma}\right)^{-1/\xi}\right) & \text{if } 1+\frac{\xi(x-\mu)}{\sigma} \ge 0\\ \exp\left(-\exp\left(-\exp\left(-\frac{(x-\mu)}{\sigma}\right)\right) & \text{if } x \in \mathbb{R}, \xi = 0 \end{array} \right\},$$
(43)

which, after differencing $H_{\mu,\sigma,\xi}(x)' = \frac{d}{dx}H_{\mu,\sigma,\xi}(x)$, yields the probability density function

$$h_{\mu,\sigma,\xi}(x) = \frac{1}{\sigma} (1 + \xi \frac{(x-\mu)}{\sigma})^{(-1/\xi)-1} exp(-(1+\xi \frac{(x-\mu)}{\sigma})^{-1/\xi}).$$
(44)

Thus, the estimated j^{th} univariate marginal density function of each expected loss series (bank-bybank) converging to GEV in the limit is defined as

$$\hat{y}_j(x) = \left(1 + \hat{\xi}_j \frac{(x - \mu_j)}{\hat{\sigma}_j}\right)_+^{-1/\xi} > 0 \text{ (for } j = 1, ..., m), \tag{45}$$

where $1 + \hat{\xi}_j(x - \mu_j)/\sigma_j > 0$, scale parameter $\sigma_j > 0$, location parameter μ_j , and shape parameter $\hat{\xi}_j \neq 0$. These moments are estimated concurrently by means of ML, which identifies possible limiting

²¹Any moments of order greater than $1/\xi$ are unbounded, i.e., $1/\xi$ indicates the highest bounded moment for the distribution. This is of practical importance since many results for asset pricing in finance rely on the existence of several moments. When the index is equal to zero, the distribution H corresponds to a Gumbel type. When the index is positive, it corresponds to a Fréchet distribution; when the index is negative, it corresponds to Weibull distribution. The Fréchet variant of modeling the asymptotic tail behavior has been found to be most appropriate for financial sector data exhibiting extremes.

laws of asymptotic tail behaviour, i.e., the likelihood of even larger extremes converges to zero as the level of statistical confidence approaches certainty. The ML estimator is evaluated numerically to solve for GEV parameters by using an iterative procedure (e.g., over a rolling window of τ observations with periodic updating). The ML estimator is evaluated numerically to solve for the GEV parameters by using an iterative procedure (e.g., over a rolling window of τ observations with periodic updating). The ML estimator is evaluated numerically to solve for the GEV parameters by using an iterative procedure (e.g., over a rolling window of τ observations with periodic updating) to maximise the likelihood $\prod_{i=1}^{n} h_{\mu,\sigma,\xi}(x \mid \theta)$ over all three parameters $\theta = (\mu, \sigma, \xi)$ simultaneously, using the linear combination of ratios of spacings estimator (Jobst and Gray (2013); Jobst (2013)).

5.2 Estimating the dependence structure of individual expected losses

We follow Jobst (2014) in defining a non-parametric, multivariate dependence function between the marginal distribution of expected losses by expanding the bivariate logistic method proposed by Pickands (1981) and to the multivariate case and adjusting the margins according to Hall and Tajvidi (2000) so that

$$Y(\omega) = \min\left\{1, \max\left\{n\left(\sum_{i=1}^{n} \Lambda_{j=1}^{m} \frac{y_{i,j}/\hat{y}_{\bullet j}}{\omega_{j}}\right), \omega, 1-\omega\right\}\right\},\tag{46}$$

where $\hat{y}_{\bullet j} = \sum_{j=1}^{n} y_{i,j}/n$ reflects the average marginal density of all $i \in n$ put option values and $0 \leq \max(\omega_1, ..., \omega_{m-1}) \leq Y(\omega_j) \leq 1$ for all $0 \leq \omega_j \leq 1$. Y(.) represents a convex function on [0, 1] with Y(0) = Y(0) = 1, i.e., the upper and lower limits of Y(.) are obtained under complete dependence and mutual independence, respectively. It is estimated iteratively (over half-year rolling windows) subject to the optimisation of the (m-1)-dimensional unit simplex

$$S_m = \left\{ (\omega_{1,\dots,}\omega_{m-1}) \in^m_+: \omega_j \ge 0, 1 \le j \le m-1; \sum_{j=1}^{m-1} \omega_j \le 1 \text{ and } \omega_m = 1 - \sum_{j=1}^{m-1} \omega_j \right\}, \quad (47)$$

which establishes the degree of coincidence of multiple series of cross-classified random variables, similar to a χ -statistic that measures the statistical likelihood that observed values differ from their expected distribution (Jobst and Kamil (2008)). This specification stands in contrast to a general copula function that links the marginal distributions using only a single (and time-invariant) dependence parameter.

5.3 Estimating the joint distribution of expected losses

We then combine the marginal distributions of these individual expected losses with their dependence structure to generate a multivariate generalised extreme value distribution (MGEV) over the same estimation period as above. Analogous to equations (43)-(45), the resultant cumulative distribution function is specified as

$$G_{t,m}(x) = \exp\left\{-\left(\sum_{j=1} y_{t,j}\right) Y_t(\omega)\right\}$$
(48)

with corresponding probability density function

$$g_{t,m}(x) = \widehat{\sigma}_{t,m}^{-1} \left(\left(\sum_{j=1}^{m} y_{t,j} \right) Y_t(\omega) \right)^{\xi_m + 1} \exp\left\{ - \left(\sum_{j=1}^{n} y_{t,j} \right) Y_t(\omega) \right\}$$
(49)

at time $t = \tau + 1$ by maximising the likelihood function $\prod_{j=1}^{n} g_{t,m}(x \mid \theta)$ over all three parameters $\theta = (\mu, \sigma, \xi)$ simultaneously. Equation (49) above represents the functional form of the joint expected losses of all firms based on the empirical observations in equation (38). Since the logarithm is a continuously increasing function over the range of the likelihood, the parameter values that maximise the likelihood will also maximise the logarithm as global maximum. Thus, we can write the lognormal likelihood function as $\sum_{j=1}^{m} \ln g_j(x \mid \theta)$ so that the ML estimate of the true values θ_0 is

$$\widehat{\theta}_{MLE} = \underset{\theta \in \Theta}{\arg \max} \widehat{l}(\theta \mid x) \to \theta_0.$$
(50)

5.4 Estimating a tail risk measure of joint expected losses

Finally, we obtain the joint expected shortfall (ES) (or conditional VaR) as a CTE measure. ES defines the probability-weighted residual density (i.e., average aggregated expected losses) beyond a pre-specified statistical confidence level ("severity threshold") over a given estimation period. It is a coherent risk measure, i.e., it satisfies several axioms of convexity, positive homogeneity, and sub-additivity (Artzner et al. (1999)); this is an improvement over VaR, which, in addition to being a pure frequency measure, is incoherent (e.g. it violates the axiom of sub-additivity). Given the dependence structure defined in equation (46), and in accordance with the general definition of ES, as

$$ES_n = E\left[X \mid X > VaR_a\right],$$

which can be written as

$$\frac{1}{1 - F\left(VaR_a\right)} \int_{VaR_a}^{\infty} \left(1 - F\left(x\right)\right) dx = \frac{1}{1 - a} \int_{a}^{1} VaR_a da$$
(51)

for a continuous random variable x, where $VaR_a = \inf \{x : F(x) \succeq a\}$ is the quantile of order 0 < a < 1 (say, a = 0.95) pertaining to the distribution function F, ES in a multivariate context is defined as

$$ES_{t,\tau,m} = E\left[z_t \mid z_t \ge G_{t,\tau,m}^{-1}\left(a\right) = VaR_{t,a}\right],\tag{52}$$

where

$$\frac{1}{a}\int_{0}^{\infty}G_{t,\tau,m}^{\leftarrow}\left(z\right)dz = \frac{1}{1-a}\int_{0}^{a}G_{t,\tau,m}^{-1}\left(a\right)da,$$

 $G^{\leftarrow}(a) = G^{-1}(a)$ is the quantile corresponding to probability 1-a, with $G^{\leftarrow}(a) \equiv \inf (z \mid G(z) \ge a))$, with random variable $z \in \mathbb{R}$, so that

$$VaR_{t,a} = \sup\left\{G_{t,\tau,m}^{-1}\left(a\right) \mid \Pr\left[z_{t} > G_{t,\tau,m}^{-1}\left(a\right)\right] \ge a\right\},$$
(53)

with the point estimate of joint potential losses of m firms defined as²²

$$G_{t,\tau,m}^{-1}\left(a\right) = \widehat{\mu}_{t,m} + \widehat{\sigma}_{t,m} / \widehat{\xi}_{t,m} \left(\left(-\frac{\ln\left(a\right)}{Y_{j}\left(\omega\right)} \right)^{-\widehat{\xi}_{t,m}} - 1 \right).$$
(54)

We also apply alternative systemic risk methodologies, such as CoVaR and SES, to the expected losses generated by equation (37) to cross-validate our main results with those obtained under alternative aggregation techniques. In their original formulation, however, both approaches specify default risk as the probability of market capitalisation to drop below a pre-specified threshold based on the historical dynamics of equity price returns rather than a measure of market-implied default risk with an observable distress barrier (such as the single-firm credit risk specification of CCA). So we need to re-formulate the definition of CoVaR and SES using expected losses as a measure of market-implied default risk. In the case of the former, we also disregard controlling for the median CoVaR (which is used to derive the marginal contribution of each bank in the form of $\triangle CoVaR$), and in the case of the latter, we derive the marginal expected shortfall (MES) of each bank (without controlling for leverage and market capitalisation) to derive a system-wide measure.

For CoVaR (conditional on individual stress of any bank), we can write

$$CoVaR_{t,\tau,a} = \sup\left\{P_{t,\tau}^{-1}(a) \mid (\Pr\left[x_t > P_{t,\alpha}^{-1}(a)\right] \ge a)(1 + \prod_{j}^{m} \Pr\left[y_{t,j} > Q_{t,\tau,j}^{-1}(a)\right] \ge a)\right\}, \quad (55)$$

where $P(x) = X_m^n$ and $Q(y) = X_j^n$ define the distribution functions of the vector-valued series of expected losses of all sample banks (as per equation (24)) and those of each individual bank over the study period, respectively.

For SES (which is based on CTE), we have

$$SES_{t,a,m} = \sum_{j}^{m} w_{t,\tau,j} \sup \left\{ Q_{t,\tau,j}^{-1}(a) \mid E\left[x_t \mid x_t \ge P_{t,\tau}^{-1}(a) = VaR_{t,a,m}\right] \right\},$$
(56)

where

$$VaR_{t,a,m} = \sup\{P_{t,\tau}^{-1}(a) \mid \Pr[x_t > P_{t,\tau}^{-1}(a)] \ge a\}.$$

The marginal expected shortfall (MES) corresponds to the partial derivative of SES with respect to the weight w of bank j within the total sample, so that $MES_{t,\alpha,j} = \frac{\partial SES_{t,\alpha,m}}{\partial w_{t,\tau,j}}$.

Since we consider all sample banks to comprise the domain of systemic risk, we simplify equation (56) above to

$$MES_{t,a,j} = \sup \left\{ y_{t,j} \mid E\left[x_t \mid x_t \ge P_{t,\tau}^{-1}\left(a\right) = VaR_{t,a,m}\right] \right\}.$$
(57)

We keep the statistical confidence a = 0.95 consistent with the one used for the Systemic CCA model and maintain the same estimation period of rolling windows with daily updating. Given that

 $^{^{22}}$ This estimation has been fully implemented in the R and C# programming languages (with all relevant codes deposited as Github under https://github.com/yuxuzi/macroprudential), and is available as web-based tool at https://macroprudential.shinyapps.io/SystemCCA/. See Jobst and Liu (2017) for further information.

the two alternative methods apply different measurement concepts of tail risk (i.e., VaR vs. CTE), we also formulate CoVaR in CTE terms as "Conditional Expected Shortfall (CoES)" so that

$$CoES_{t,a,m} = E\left[x_t \mid x_t \ge P_{t,\tau,j}^{-1}\left(a\right) = VaR_{t,a,j}\right] \cap E\left[y_{t,j} \mid y_{t,j} \ge Q_{t,\tau,j}^{-1}\left(a\right) = VaR_{t,a,j}\right]$$
(58)

and MES in VaR terms as "MarginalVaR (MVaR)" so that

$$\sum_{j}^{m} MVaR_{t,a,j} = \sum_{j}^{m} \sup\left\{ y_{t,j} \mid P_r\left[x_t > P_{t,\tau}^{-1}\left(a\right)\right] \ge a \right\}.$$
(59)

5.5 Empirical Application and Findings

Based on the specification of default risk on a system-wide level in the previous section, we estimate the multivariate probability distribution of joint expected losses for all sample banks. We combine the univariate density function of estimated individual expected losses and their non-parametric dependence structure (over a 120-day sliding window and daily updates) under the assumption of GEV-consistent asymptotic tail behaviour according to equation (48). With a fully-specified multivariate probability distribution, we can then extract point estimates of expected losses at varying levels of statistical confidence at different points in time by inverting the cumulative density function of joint expected losses (see equation 52). In order to generate a coherent risk measure of systemic risk for tail events, we focus on expected shortfall (ES) according to equation (53). Table 1 provides an overview of all estimation results at different levels of statistical confidence over the total study period (since 2005) and during episodes of significant changes in macroeconomic or market conditions. Figure 7 shows in more detail the daily time series of the sum of expected losses of each bank and the median value of the joint expected losses.

We find that the systemic risk of the largest U.K. banks has greatly diminished in the wake of the global financial crisis. During the crisis, the sharp decline in banks' asset values raised the 95% CTE of joint expected losses to levels that would have completely exhausted the total available amount of system-wide equity (GBP187.5 billion). However, since 2012, the 95% CTE dropped sharply to only GBP 38.8 billion (or slightly more than one-third of the pre-crisis value (GBP 111.5 billion), which implies a far lower probability of banks experiencing extreme losses at the same time (see Table 1).

Accounting for the occurrence of large and interrelated changes of expected losses provides an early indication of systemic risk outside known episodes of system-wide stress (such as the global financial crisis and the European debt crisis). We find long episodes of elevated joint expected losses, which are not captured by an identification strategy of systemic risk that follows a simple aggregation of individual expected losses (without considering their distributional properties and dependence accross sample banks). The pre-crisis 95% CTE of joint expected losses exceeded the sum of individual expected losses by almost a factor of three (see Table1). During times of stress (such as the global financial crisis), joint expected losses become very sensitive to extreme shocks at higher levels of statistical confidence, indicating high elasticity of systemic risk (with a certain degree of persistence). These (in-sample) results suggest some early warning capacity of our approach in identifying the stochastic relevance and persistence of changes in the way expected losses evolve over time, both individually and jointly. This is also consistent with Kerry (2019), who finds that equity market-based capital ratios Table 1: UK Banking Sector (Largest Listed Banks): Sum of individual and joint Expected Losses at Different Levels of Statistical Confidence, 2005-2017 (In GBP Billions)

Table 1. UK Banking Sector (Largest Listed Banks): Sum of Individual and Joint Expected Losses at Different Levels of Statistical Confidence, 2005-2017 (In GBP Billions)							
	Pre-Financial Crisis: June 2005– Aug. 2008	Global Financial Crisis: Aug. 2008– March 2010	Sovereign Debt Crisis: April 2010– Dec. 2011	Post-Crisis Jan. 2012– Dec. 2017	Entire Time Period		
	Joint Expected Losses						
	Conditional Tail Expectation (95%)						
25th percentile	1.2	93.4	1.3	0.5	0.9		
Median	45.1	116.0	7.2	0.9	3.0		
75 th percentile	76.3	151.4	27.9	2.9	62.9		
Average	111.5	266.2	106.2***	38.8***	119.1		
	Value-at-Risk (95%)						
25 th percentile	1.0	92.1	1.3	0.5	0.8		
Median	28.1	104.7	4.1	0.9	2.2		
75 th percentile	57.3	146.2	22.8	2.4*	34.0		
Average	36.2	187.2	55.3***	27.8***	50.9		
	Median						
25 th percentile	0.4	14.4	0.7	0.3	0.3		
Median	0.7	80.7	1.3	0.5	0.7		
75 th percentile	42.4	90.5	3.4	0.8	14.2		
Average	18.9	65.0	4.5	1.7	14.3		
	Sum of Individual Expected Losses						
25 th percentile	0.5	18.9	1.5	0.6	0.6		
Median	30.5	141.7	2.3	1.1	1.5		
75 th percentile	76.1	237.2	27.4	2.3*	25.6		
Average	49.8	139.2	14.9	1.6	33.0		

Sources: Bloomberg L.P. and authors. Note: The estimation window of 120 working days and daily updating for the multivariate distribution of joint expected losses means that the output time series of the estimation results starts on 21 June 2005 (based on a sample start date of 1 January 2005). The statistical significance of expected losses (for each time period) is determined via a simple two-sided t-test (based on the distribution of the sum of individual expected losses forming the expectation specific to each time period): ***=1 percent, **=5 percent, *=10 percent statistical significance.

Figure 7: UK Banking Sector (Largest Listed Banks): Sum of Individual and Joint Expected Losses, 2005-17 (In GBP Billions) 1/



Figure 7. UK Banking Sector (Largest Listed Banks): Sum of Individual and Joint Expected Losses, 2005-17 (In GBP Billions) 1/

Sources: Bank of England, Bloomberg L.P., and authors. Note: 1/ shown as 120-day moving average.

would have been better at signalling bank distress in the run-up to the global financial crisis than regulatory capital ratios - particularly the Tier 1 capital ratio.

Our portfolio-based estimation approach seems to also capture considerable diversification effects due to banks' varying sensitivity to price shocks. The dependence structure of expected losses has a significant impact on the estimated magnitude of systemic risk over the study period. We find that the median of joint expected losses is less than half the sum of the individual expected losses (GBP 0.7 billion vs. GBP 1.5 billion), suggesting considerable variability in the way financial stress propagates across sample banks. The difference remains persistent but declines somewhat during stress periods (indicating high co-movement of expected losses) when the dependence of expected losses tends to increase. The parameterisation of the multivariate GEV distribution of joint expected losses according to equation (49) suggests that the average dependence value across all sample banks peaked during the global financial crisis at a value of 0.57 (up from 0.47 during the pre-crisis period) but has remained relatively stable since at 0.56 until the end of the study period in 2017 (see Table 4).

Alternative systemic risk measures deliver similar results if adapted to our panel series of expected losses but leave large gaps outside severe distress periods. The CoVaR and SES measures of joint expected losses at 95% statistical confidence converge to those generated using the Systemic CCA framework, with median values during the global financial crisis at GBP 140.8 billion and GBP 148.6 billion, respectively (see Table 2).²³However, the bivariate specification of both approaches fails to fully capture very large changes in systemic risk over time, which does not only generate discontinuities in measuring systemic risk over time but also diminishes their early warning capacity.

Our results help address the challenges from the inherent uncertainty and time-inconsistency in measuring systemic risk - the magnitude of risk is imperfectly predictable, and its impact cannot be precisely quantified in advance, which requires flexible approaches that can capture a broad range of outcomes at varying levels of statistical confidence. However, we acknowledge the limitations of our approach (and are reflected as caveats in the interpretation of the findings). The underlying option pricing model as specified in equation (5) (given its specific distributional assumptions, the derivation of both implied assets and asset volatility and assumptions about the default barrier) could fail to capture some relevant economics underpinning default risk and, thus, could generate biased estimates of expected losses (even if we incorporate jump risk). Moreover, financial market behaviour, especially during times of stress, might defy statistical assumptions of valuation models, such as the option pricing theory and extreme value measurement. For instance, during the global financial crisis and the European sovereign debt crisis, financial market behaviour was characterised by rare and non-recurring events, and not by repeated realisations of predictable outcomes, generated by a stochastically stable process of asset price changes. Thus, sudden and unexpected realisations beyond historical precedent defy the statistical apparatus underlying conventional asset pricing theory even if we account for tail events. Also, equity prices might not only reflect fundamental values due to both shareholder dilution and trading behaviour that can obfuscate proper economic interpretation, during periods of stress, such as the global financial crisis.

 $^{^{23}}$ Note that the average values of joint expected losses under Systemic CCA are sometimes lower than those derived from conditional measures, such as CoVaR and SES, outside of stress periods; this can be attributed to the distributional approach of Systemic CCA, which generates continuous point estimates of joint expected losses while actual expected losses might not have breached the stochastic threshold.

Table 2: UK Banking Sector (Largest Listed Banks): Joint Expected Losses - Comparison of Systemic Risk Methodologies, 2005-2017

Table 2. UK Banking 2005-2017	Sector (Largest Listed (In GBP Billions)	Banks): Joint Expected L	osses—Comparison of Sy	stemic Risk Method	dologies,		
	Pre-Financial Crisis: June 2005– Aug. 2008	Global Financial Crisis: Aug. 2008– March 2010	Sovereign Debt Crisis: April 2010– Dec. 2011	Post-Crisis Jan. 2012– Dec. 2017	Entire Time Period		
			Systemic CCA				
		Joint Expected Los	sses: Conditional Tail Expectation	on (95%)			
Median	45.1	116.0	7.2	0.9	3.0		
Average	111.5	266.2	106.2	38.8	99.5		
	Joint Expected Losses: Value-at-Risk (95%)						
Median	28.1	104.7	4.1	0.9	2.2		
Average	36.2	187.2	55.3	27.8	54.3		
	CoVaR Methodology						
	Conditional Expected Shortfall (CoES) at 95% ^{1,2,3}						
Median	118.5	143.6	38.1	0.2	27.3		
Average	131.0	134.2	63.8	10.8	59.1		
	Conditional Value-at-Risk (CoVaR) at 95%						
Median	115.6	140.8	37.5	0.2	26.1		
Average	121.8	121.1	60.1	9.9	54.8		
	SES Methodology						
	Systemic Expected Shortfall (SES) at 95% ¹						
Median	119.4	148.6	39.4	0.3	30.4		
Average	139.0	142.4	67.1	11.5	62.7		
		Systemic Value-at-Risk (SVaR) at 95%					
Median	118.3	163.0	38.9	0.2	30.9		
Average	134.7	133.7	67.1	11.3	61.1		

Sources: Bloomberg L.P. and authors. *Note:* 1/ <u>Systemic Expected Shortfall</u>; SES aggregates (as weighted sum) the individual losses of all institutions ("marginal expected shortfall" or MES) conditional on a systemic stress event (i.e., the banking sector experiencing losses above a level consistent with its conditional tail expectation (CTE) of 95 percent, estimated over a rolling window of 120 working days) – for a consistent comparison of SES based on CCA-generated expected losses with results obtained under the CoVaR methodology, a systemic stress level of VaR at 95 percent, was chosen as an alternative threshold (which turns SES Into what could be termed "Systemic VaR (Systemic VaR (Systemic VaR (Systemic VaR (Systemic VaR) (Systemic VaR)

6 Integrated Market-Implied Capital Assessment

The historical estimates of individual and joint expected losses also motivate an integrated marketbased capital assessment using the risk-adjusted balance sheet of each bank. We can replicate two conventional measures of solvency risk - the regulatory capital adequacy and default probability using key variables that are fully indentified and estimated within our CCA framework (see Section 3): (i) the market-implied capital adequacy ratio (MCAR) between the market capitalisation and the implied asset value as well as (ii) the fair value credit spread (based on the ratio between expected losses and the default barrier). All variables of these two measures can be extracted from equation (29), which specifies expected losses as a put option on the bank's implied asset value subject to a pre-determined debt repayment schedule that defines the default barrier. The fair value credit spread can be obtained by rearranging the equation for a given level of expected losses. Figure 8 shows the non-linear relation between the average MCAR and the average credit spread consistent with average (individual) expected losses (that is, without considering the dependence structure between expected losses). According to equation (52) above, a lower market capitalisation implies a lower asset value to outstanding liabilities, which reduces the MCAR due to higher default risk (and increases asset volatility). Similarly, higher expected losses increases the fair value credit spread. The non-linear interaction between these variables can also be shown based on the joint expected losses and the corresponding MCAR and credit spread (consistent with their average elasticity across all sample banks). The empirical results suggest a generally lower sensitivity of both MCAR and credit spreads to changes in default risk if the joint distribution of expected issues is considered (see Figure 8).²⁴

 $^{^{24}}$ Aikman et al. (2019) indirectly confirm the complementarity of accounting and market-based measures of solvency risk. They find that even in hindsight it would have been hard to diagnose fully the risks in the run-up to the crisis. This suggests that macroprudential policy frameworks should be calibrated with some built-in "slack" to account for the inherent difficulty of risk assessment, particularly in real time.

Figure 8: UK Banking Sector (Largest Listed Banks): Integrated Capital Assessment Using Contingent Claims Analysis (CCA) - Market-Implied Capital Adequacy Ratio (MCAR), Expected Loss Ratio, and Fair Value Credit Spread





Sources: Bank of England, Bloomberg L.P., and authors. Note: The two charts show the non-linear relation between three interlinked elements of the CCA framework—the market-implied capital adequacy ratio (MCAR), the expected loss (EL), and the fair value credit spread (derived from the relation between expected losses and the default barrier). The black and red dots illustrate this integrated assessment using the average of individual (bank-by-bank) expected losses (Bank and the median value) of their multivariate distribution (derived from the agregate agregated losses and median joint expected losses to the median market capitalisation across all sample banks. Store the MCAR and the credit systemic CAI). The EL ratio scales the average expected losses and median joint expected losses to the median market capitalisation across all sample banks. Store the MCAR and the credit systemic CAI). The EL ratio scales the the median value of joint expected losses using () the elasticity of the average expected loss in GBP billions across all sample banks. Store the MCAR and the credit systemic ACI). The EL ratio scales the median value of joint expected losses to the average expected loss in GBP billions across all sample banks. Store the scale agregated accession of the overage expected loss in GBP billions across all sample banks. Store the scale agregated bases using () the exercise of 0.107x²/2-6.3676x(-0.9552), where x-average expected loss in GBP billions across all sample banks. Store the daverage the spected loss in GBP billions across all sample banks. Store the daverage expected loss in GBP billions across all sample banks. Store the daverage expected loss in GBP billions across all sample banks. Store the daverage expected loss in GBP billions across all sample banks over the study period.

The MCAR represents a market-based analogue to the accounting-based metrics of regulatory capital adequacy reported by the sample banks, and indicates whether the default risk implied by prudential measures is consistent with individually and jointly estimated expected losses inferred from market prices using the CCA. We compare the MCAR to the core leverage ratio to examine the difference between the economic and book value of both assets and capital (see Figure 9). The core leverage ratio is a simple indicator of solvency.²⁵ It relates capital resources of a bank to the size of its balance sheet exposures. But it does not take account of the riskiness of exposures like the risk-weighted capital requirements. We find that there is a persistent gap between both measures over the study period, which varies over time and declines during times of stress. Thus, the leverage ratio acts as an important guardrail against model risks embedded in risk-weighted measures of exposures and may be viewed as a stabiliser against excessive balance sheet expansion funded by debt. It is readily understood by market participants and more comparable across firms than risk-weighted measures. The average core leverage ratio of the four sample banks has roughly doubled since 2009.

Similarly, relating the MCAR to the risk-based capital requirement generates a structural modelbased representation of the bank's price-to-book (PB) ratio (see Figure 10).²⁶ We choose the capital ratio between common equity Tier 1 capital and risk-weighted assets ("Common Equity Tier 1 (CET1) capital ratio") as a measure of capital adequacy, which is conceptually closest to the composition of MCAR. In normal times, we would expect the PB ratio to be greater than one, because it not only reflects the bank's current net asset value but also includes the present value of future profits. However, in the aftermath of the global financial crisis, the average PB ratio of the sample banks dropped well below one as the regulatory capital ratio and the MCAR have begun to diverge. This is consistent with the market taking a more pessimistic view about banks' future profitability (relative to its cost of equity) than was the case prior to the crisis, with the MCAR as a "shadow capital ratio," reflecting the actual cost of equity.²⁷ Forces driving prices and risk are different in a crisis than they would be out of crisis. ²⁸ Prior to the crisis, the MCAR was considerably higher than the levels seen today despite banks holding much lower capital, which might be explained by markets having significantly under-priced risk previously.

PB ratios improved again in 2012 as equity markets recovered and banks began further strengthening capital buffers. The amount of CET1 capital that the sample banks now hold relative to their risk-weighted assets, via their minimum capital requirements and various buffers, has increased almost tenfold since the global financial crisis (Bank of England (2018)). Figure 10 shows that the sample banks' PB ratios have recovered from their intermittent trough in mid-2016 and seem to have stabilised around 0.9.

 $^{^{25}\}mathrm{This}$ is defined here as shareholders equity divided by (accounting) balance sheet assets.

²⁶A bank's PB ratio compares the market value of shareholders' equity in the bank with its accounting or book value. In theory, there are at least two reasons why companies should have a low PB ratio: investors are unsure that a company's net assets are correctly valued, or it has low or highly uncertain future profit prospects. The UK Financial Policy Committee has judged that UK banks' low PB ratios could be explained by market concerns over expected future profitability rather than by concerns about asset quality.

²⁷This gap arises from differences between (i) the market valuation of banks vis-ā-vis accounting valuation of their assets and/or (ii) risk-weighted assets vis-a-vis total assets, which are influenced by (a) structural (crisis vs. post-crisis), (b) cyclical (risk aversion), and (c) technical (prudential/regulatory standards) factors.

 $^{^{28}}$ Danielsson (2002) argues that since market data is endogenous to market behaviour, statistical analysis made in times of stability does not provide much guidance in times of crisis. The distribution of risk is different during crisis than in other periods and this complicates the role of regulation.

Figure 9: UK Banking Sector (Largest Listed Banks): Accounting-Based and Market-Implied Capital Adequacy of All Sample Banks, 2005-17 (In Percent)





Sources:: Bank of England, Bloomberg L.P., and authors; Note: MCAR = market-implied capital adequacy ratio (which is defined as market capitalisation divided by option-based implied asset value); 1/ The unconditional MCAR is based on the average of each sample bank's individual asset value implied by expected losses; 2/ The conditional MCAR is based on the (theoretical) implied asset value that corresponds to the median joint expected losses consistent with the historical relationship between implied asset values and expected losses of all sample banks over the study period; 3/ The core leverage ratio does not control for risky assets (which are deducted in the prudential leverage ratio) and also tends to be more volatile since there is no netting of derivatives assets and liabilities (and other claims/obligations that have expanded significantly during the global financial crisis).

Figure 10: UK Banking Sector (Largest Listed Banks): Difference between Accounting-Based and Market-Implied Capital Adequacy for All Sample Banks, 2005-17 (In Percent)





Sources: Bank of England, Bloomberg L.P., and authors. Note: smoothed line shows 120-working day moving average of the black line; CET1=Common Equity Tier 1; MCAR=market-implied capital adequacy ratio of market capitalisation divided by option-based implied asset value otted line shows the 120-working day moving average. The market perception of bank solvency implies higher default risk than implied by accounting-based measures if the CET1 capital ratio exceeds MCAR.

7 Conclusion

This paper applied the Systemic CCA framework (Jobst and Gray (2013)) to model the marketimplied systemic solvency risk in the U.K banking sector by controlling for common factors affecting the interlinkages between individual risk-adjusted balance sheets of the largest commercial banks. This approach explicitly acknowledges nonlinearities in measuring default risk using a GEV distribution setup to quantify simultaneous distress. The underlying option pricing formula for measuring expected losses was augmented with a jump diffusion process of banks' asset values to mitigate the empirical shortcomings of traditional single-firm structural default models.

Accounting for the occurrence of large and interrelated changes of expected losses provides an early indication of systemic risk outside known episodes of system-wide stress (such as the global financial crisis and the European debt crisis). We find long time periods of elevated joint expected losses, which are not captured by an identification strategy of systemic risk that follows a simple aggregation of individual expected losses (without considering their distributional properties and dependence across sample banks). During times of stress (such as the global financial crisis), joint expected losses become very sensitive to extreme shocks at higher levels of statistical confidence. The cross-validation of our results with the aggregate outcomes of alternative (bivariate) systemic risk measures, such as CoVaR and SES, over the same study period suggests a greater early warning capacity of our approach.

We also exploited the integrated way of measuring expected losses to generate a market-implied measure of capital adequacy, the MCAR. This measure of "shadow capital adequacy" indicated whether estimated expected losses - individually and jointly - were consistent with the default risk implied by the accounting-based capital adequacy ratio reported by sample banks. Hence, this distribution-based perspective of market-implied solvency risk can inform a system-wide capital adequacy assessment that reflects the variability of both assets and liabilities at different levels of statistical confidence.

Market-based measures of systemic risk can effectively complement prudential reporting in informing a more comprehensive assessment of capital adequacy by considering the impact of changes in market conditions on the perceived risk profile of banks. This has become increasingly relevant due to fundamental changes in the market assessment of banks' solvency risk following the global financial crisis. Prior to the crisis markets underpriced the risk inherent in debt capital which enabled banks to fund themselves at rates that did not reflect the risk associated with such investments. Investors are now more uncertain about the value of banks' net assets and of the underlying asset risks. Low market values may also reflect weak or uncertain profits, or high equity risk premia. The contribution of each factor is not entirely independent, and will vary by bank.

Going forward, further refinements to our model are possible, including various simulation approaches, which might come at the expense of losing analytical tractability. The current specification could also be expanded in scale and scope to include different financial institutions and other types of risks. For instance, given the flexibility of this framework, the financial sector and sovereign risk analysis could be integrated with macro-financial feedback effects to design monetary and fiscal policies. Such an approach could inform the calibration of stress scenarios and sovereign balance sheets and the appropriate use of macroprudential regulation.

References

- Acharya, Viral V., Lasse H. Pedersen, Tomas. Philippon, and Matthew Richardson, "Measuring Systemic Risk," Working Paper 1002, Federal Reserve Bank of Cleveland 2010.
- Adrian, Tobias and Markus K. Brunnermeier, "CoVaR," American Economic Review, October 2016, 106(7), 1705–41.
- Aikman, David, Jonathan Bridges, Anil Kashyap, and Casper Siegert, "Would Macroprudential Regulation Have Prevented the Last Crisis," *Journal of Economic Perspectives*, 2019, 33 (1), 107–30.
- Allen, Franklin, Ana Babus, and Elena Carletti, "Financial Connections and Systemic Risk," Technical Report, NBER 2010.
- Artzner, Philippe, Freddy Delbaen, Jean-Marc Eber, and David Heath, "Coherent Measures of Risk," *Mathematical Finance*, 1999, 9, 203–28.
- Ball, Clifford A. and Walter N. Torous, "A Simplified Jump Process for Common Stock Returns," Journal of Financial and Quantitative Analysis, 1983, 18(1), 53–65.
- Banelescu, Georgiana-Denisa and Elena-Ivona Dumitrescu, "What at the SIFIs: A Component Expected Shortfall (CES) Approach to Systemic Risk," EUI Working Paper MWP 2013/23, San Domenico di Fiesole: European University Institute August 2013.
- Bank of England, "Financial Stability Report," Technical Report Issue No. 43, June 2018.
- Bardoscia, Marco, Paolo Barruca, Adam Brinley Codd, and John Hill, "The Decline of Solvency Contagion Risk," Working Paper 662, Bank of England, June 2017.
- Bernardi, Mauro, Antonella Maruotti, and Lea Petrella, "Multiple Risk Measures for Multivariate Dynamic Heavy-tailed Models," *Journal of Emprical Finance*, 2017, 43 (C), 1–32.
- Bisias, Dimitrios, Mark Flood, Andrew W. Lo, and Stavros Valavanis, "A Survey of Systemic Risk Analytics," Annual Review of Financial Economics, 2012, 4 (1), 255–96.
- Black, Fisher and Myron S. Scholes, "The Pricing of Options and Corporate Liabilities," *Journal* of Political Economy, 1973, 81, 637–59.
- Blancher, Nicolas, Srobona Mitra, Hanan Morsy, Akira Otani, Tiago Severo, and Laura Valderrama, "Systemic Risk Monitoring ("SysMo") Toolkit A User Guide," Working Paper 13/168, IMF, Washington D.C.: International Monetary Fund 2013.
- Borwein, Jonathan M., David M. Bradley, and Richard Crandall, "Computational Strategies for the Riemann Zeta Function," *Journal of Computational and Applied Mathematics*, 2000, 121, 247–96.
- Brownlees, Christian and Robert Engle, "Volatility, Correlation and Tails for Systemic Risk Measurement," Working Paper, NYU Stern School of Business May 2011.

- Campbell, John Y., Martin Lettau, Burton G. Malkiel, and Yexiao Xu, "Have Individual Stock Become More Volatile? An Empirical Exploration of Idiosyncratic Risk," *Journal of Finance*, 2001, Vol. 56(1), 1–43.
- Cerutti, Eugenio, Stijn Claessens, and Patrick McGuire, "Systemic Risks In Global Bankng: What Available Data can tell us and What More Data are Needed?," Working Paper 11/222, IMF, Washington D.C., International Monetary Fund 2011.
- Cesare, Antonio Di and Anna Rogantini Picco, "A Survey of Systemic Risk Indicators," Occasional Papers 458, Banca D'Italia October 2018.
- Coles, Stuart, Janet Heffernen, and Jonathan Tawn, "Dependence Measures for Extreme Value Analysis," *Extremes*, 1999, 56, 339–65.
- Cox, John C., Stephen A. Ross, and Mark Rubinstein, "Option Pricing: A Simplified Approach," *Journal of Financial Economics*, 1979, 7 (3), 229–63.
- **Crosby, Peter and Jeff Bohn**, "Modeling Default Risk," Technical Report, Moody's KMV Company 2003.
- **Danielsson, Jon**, "The Emperor Has No Clothes," *Journal of Banking and Finance*, 2002, *26*, 1273–96.
- Dumas, Bernard, Jeff Fleming, and Robert E. Whaley, "Implied Volatility Functions: Empirical Tests," *Journal of Finance*, 1998, 53 (6), 2059–106.
- Elsinger, Helmut, Alfred Lehar, and Martin Summer, "Using Market Information for Banking System Risk Assessment," International Journal of Central Banking, 2006, 2 (1), 137–66.
- Fisher, Ronald A. and Leonard H.C. Tippett, "Limiting Forms of the Frequency Distribution of the Largest or Smallest Member of a Sample," *Proceedings of the Cambridge Philosophical Society*, 1928, 24, 180–90.
- **Giudici, Paolo and Laura Parisi**, "CoRisk: Measuring Systemic Risk through Default Probability Contagion," DEM Working Paper Series 116, University of Paris, Department of Economics and Management 2016.
- Gnedenko, Boris V., "Sur la Distribution Limite du Terme Maximum d'Une," Annals of Mathematics, 1943, 44, 423–53.
- Gray, Dale and Samuel Malone, Macrofinancial Risk Analysis, Wiley Finance, 2008.
- Gray, Dale F. and Andreas A. Jobst, "New Directions in Financial Sector and Sovereign Risk Management," Journal of Investment Management, 2010, 8(1), 23–8.
- and _, "Modeling Systemic Financial Sector and Sovereign Risk," Sveriges Riksbank Economic Review, 2011, 2, 68–106.

- Haldane, Andrew G., "Capital Discipline," January 2011. Speech given at American Economic Association, Denver/Colorado.
- _ and Benjamin Nelson, "Tails of the Unexpected," 2012. Paper given at "The Credit Crisis Five Years On: Unpacking the crisis", conference held at the University of Edinburgh Business School.
- Hall, Peter and Nader Tajvidi, "Distribution and Dependence-function Estimation for Bivariate Extreme Value Distributions," *Bernoulli*, 2000, *6*, 835–44.
- Hamilton, James D., Time Series Analysis, Princetown University Press, 1994.
- Hanson, Floyd B., John J. Westman, and Zongwu Zhu, "Multinomial Maximum Likelihood Estimation of Market Parameters for Stock Jump-Diffusion Models," *Contemporary Mathematics*, 2004, 351, 155–69.
- Hattori, Akio, Kentaro Kikuchi, and Uchida Yoshihiko, "A Survey of Systemic Risk Measures. Methodology and Application to the Japanese Market," Discussion Paper Series 2014 E-3, Institute for Monetary and Economic Studies (IMES), Tokyo: Bank of Japan April 2014.
- Heston, Steven L., "A Closed-Form Solution for Options and Stochastic Volatility with Applications to Bond and Currency Options," *Review of Financial Studies*, 1993, 6 (2), 327–43.
- and Saikat Nandi, "A Closed-Form GARCH Option Valuation Model," Review of Financial Studies, 2000, 13 (3), 585–625.
- Honore, Peter, "Pitfalls in Estimating Jump-Diffusion Models," Technical Report, University of Aarhus Aarhus School of Business 1998.
- International Monetary Fund, "People's Republic of China-Hong Kong Special Administrative Region: Technical Note on Stress Testing the Banking Sector," IMF Country Report 14/210, (Washington, D.C.: International Monetary Fund), 16 July 2014. 19-25.
- International Monetary Fund, "Assessing the Systemic Implications of Financial Linkages," Global Financial Stability Report Chapter 2, World Economic and Financial Surveys, (Washington, D.C.: International Monetary Fund), April 2009a.
- _, "Detecting Systemic Risk," Global Financial Stability Report Chapter 3, World Economic and Financial Surveys, (Washington, D.C.: International Monetary Fund), April 2009b.
- Jobst, Andreas A., "Operational Risk The Sting is Still in the Tail But the Possion Depends on the Dose," Journal of Operational Risk, 2007, 2(2), 1–56.
- _, "Multivariate Dependence of Implied Volatilities from Equity Options as Measure of Systemic Risk," International Review of Financial Analysis, June 2013, 28, 112–29.
- _, "Measuring Systemic Risk-Adjusted Liquidity (SRL) A Model Approach," Journal of Banking and Finance, 2014, 45, 270–87.

- and Dale F. Gray, "Systemic Contingent Claims Analysis Estimating Market-Implied Systemic Risk," Working Paper 13/54, IMF, Washington D.C.: International Monetary Fund 2013.
- _ and Guoyuan Liu, "Systemic Contingent Claims Analysis and Dynamic Dependence Structure Model–Implementation Tools in R," 2017. mimeo.
- and Herman Kamil, "Stock Market Linkages Between Latin America and the United States During 'Tail Events'," Regional Economic Outlook pp. 35f, Western Hemisphere Department, Washington, D.C.:International Monetary Fund 2008.
- Jones, Philip E., Scott P. Mason, and Eric Rosenfeld, "Contingent Claims Analysis of Corporate Capital Structure: An Empirical Investigation," *Journal of Finance*, 1984, 39, 611–27.
- Jordan, Dan J., Douglas Rice, Jacques Sanchez, and Donald H. Wort, "Explaining Bank Market-to-Book Ratios: Evidence from 2006 to 2009," *Journal of Banking and Finance*, 2011, 35(8), 2047–55.
- Kanno, Masayasu, "The Network Structure and Systematic Risk in the Global Non-life Insurance Market," *Insurance: Mathematics and Economics*, 2016, 67, 38–53.
- Karas, Marta and Witold Szczepaniak, "Towards a Generalized Measure of Systemic Risk: Systemic Turbulence Measure," in Krzysztof Jajuga, Hermann Locarek-Junge, Lucjan T. Orlowski, and Karsten Staehr, eds., Contemporary Trends and Challenges in Finance, Springer Proceedings in Business and Economics Cham: Springer 2019, pp. 11–21.
- Karimalis, Emmanouil N. and Nikos K. Nomikos, "Measuring Systemic Risk in the European Bankng Sector: A Copula CoVaR Approach," *European Journal of Finance*, 2018, 24 (11), 944–75.
- Kerry, Will, "Finding the Bad Apples in the Barrel: Using the Market Value of Equity to Signal Banking Sector Vulnerabilities," Working Paper 19/20, IMF, Washington, D.C. : International Monetary Fund 2019.
- Kiefer, Nicholas M., "Discrete Parameter Variation: Efficient Estimation of a Switching Regression Model," *Econometrica*, 1978, 46, 427–34.
- Killick, Rebecca, Paul Fearnhead, and Idris A. Eckley, "Optimal Detection of Change Points with Linear Computational Cost," *Journal of the Americal Statistical Association*, 2012, 107, 1590– 98.
- Lavielle, Marc, "Using Penalized Contrasts for the Change-Point Problem," Signal Processing, 2005, 85, 1501–10.
- _, "Detection of Change Points in a Time Series Statistics in Action with R," March 2017. http://sia.webpopix.org/changePoints.html.
- Law, Daniel and Shaun K. Roache, "Assessing Default Risks for Chinese Firms: A Lost Cause?," Working Paper 15/140, IMF, Washington D.C.: International Monetary Fund 2015.

- Leland, Hayne E., "Predictions of Default Probabilities in Structural Models of Debt," in H. Gifford Fong, ed., The Credit Market Handbook, John Wiley & Sons, 2006.
- Liao, Shuyu, Elvira Sojli, and Wing W. Tham, "Managing Systemic Risk in the Netherlands," International Review of Economics and Finance, 2015, 40, 231–45.
- Merton, Robert C., "On the Pricing of Corporate Debt: The Risk Structure of Interest Rates," *The Journal of Finance*, 1974, 29(2), 449–70.
- ____, "Option Pricing When Underlying Stock Returns are Continuous," Journal of Financial Economics, 1976, 3, 125–44.
- Novickyte, Lina and Victorija Dicpinigatiene, "Application of Systemic Risk Measurement Methods: A Systematic Review and Meta-analysis Using a Network Approach," *Quantitative Finance and Economics*, 2018, 2 (4), 798–820.
- **Pickands, James III**, "Multivariate Extreme Value Distributions," Proceedings of the 43rd Session of the International Statistical Institute, 1981, 49, 859–78.
- Poon, Ser-Huan, Michael Rockinger, and Jonathan Tawn, "Extreme Value Dependence in Financial Markets:Diagnostics, Models and Financial Implications," *Review of Financial Studies*, 2003, 17 (2), 581–610.
- Popescu, Alexandra and Carmelia Turcu, "Systemic Sovereign Risk in Europe: An MES and CES Approach," *Review D'economie Politique*, 2014, 124 (6), 899–925.
- Reboredo, Juan C. and Andrea Ugolini, "Systemic Risk In European Sovereign Debt Markets: A CoVaR-Copula Approach," *Journal of International Money and Finance*, 2015, 51, 214–44.
- and _ , "Systemic Risk of Spanish Listed Bank: A Vine Copula CoVaR Approach [Riesgo Sistemico de Los Bancos Espanoles Cotizados: Una Approximacion CoVaR Con Copulas Vine]," Spanish Journal of Finance and Accounting/ Revista Espanola de Financiacian y Contabilidad, 2016, 45 (1), 1-31.
- Sheu, Her-Jiun and Chien-Ling Cheng, "Systemic Risk in Taiwan Stock Market," Journal of Business Economics and Management, 2012, 13 (6), 895–914.
- Sklar, Abe, "Fonctions de Repartition et Dimensions et Leurs Marges," Publ. Inst. Statist. Univ. Paris, 1959, 8, 229–31.
- Sondow, Jonathan, "An Antisymmetric Formula for Euler's Constant," Mathematics Magazine, 1998, No. 71, 219–20.
- Synowiec, Daniel, "Jump-diffusion Models with Constant Parameters for Financial Log-return Processes," Computers and Mathematics with Applications, 2008, 56, 2120–27.
- Webber, Lewis and Matthew Willison, "Systemic Capital Requirements," Working Paper 436, Bank of England 2011.

Zhou, Chunsheng, "A Jump-Diffusion Approach to Modeling Credit Risk and Valuing Defaultable Securities," Finance and Economics Discussion Series Paper No. 15, Board of Governors of the Federal Reserve System, 1997.

Appendix A - Parameter Variation in the Diffusion Volatility (σ)

Bank 1 Bank 2 0.09 0.08 - Diffusion Volatility · Lower Bound (95% Confidence Interval)* – Diffusion Volatility -- Lower Bound (95% Confidence Interval)* -- Upper Bound (95% Confidence Interval)* 0.08 0.07 Upper Bound (95% Confidence Interval)* 0.07 0.06 0.06 0.05 0.05 0.04 0.04 0.03 0.03 0.02 0.02 0.01 0.01 0 n .05 Bank 4 Bank 3 0.2 0.07 — Diffusion Volatility
 … Lower Bound (95% Confidence Interval)*
 … Upper Bound (95% Confidence Interval)* - Diffusion Volatility 0.18 Lower Bound (95% Confidence Interval)*
 Upper Bound (95% Confidence Interval)* 0.06 0.16 0.05 0.14 0.12 0.04 0.1 0.03 0.08 0.06 0.02 0.04 0.01 0.02 (20000 04.042 Ś ŝ

Figure 11: Parameter Estimates: Variation of Parameter Estimates: Diffusion Volatility

Figure 11. Variation of Parameter Estimates: Diffusion Volatility

Source: authors' estimates. Note: */ The average standard errors over the total study period were 0.0013 (Bank 1), 0.0012 (Bank 2), 0.0021 (Bank 3), and 0.0015 (Bank 4).

Appendix B - Specification of Moments of the GEV Distribution

Since all raw moments of $G_{t,m}(\cdot)$ are defined contingent on the tail shape, the natural estimator of is $\hat{\xi}$ derived by means of the *linear combination of ratios of spacings* (LRS) method using the linear combination

$$\begin{split} \xi &= (n/4)^{-1} \sum_{i=1}^{n/4} ((ln(v_i))/(-ln(c))) \\ \text{for } n \text{ observations, where} \\ \hat{v_i} &= \frac{x_{n(1-\alpha):n} - x_{n\alpha}c_{:n}}{x_{n\alpha}c_{:n} - x_{n\alpha:n}} \text{ and } c = \sqrt{\frac{ln(1-\alpha)}{ln(\alpha)}} \text{ for quantile } \alpha = i/n. \text{ Since } x_{n\alpha:n} = G^{-1}(\alpha), \text{ the approximation } \hat{v_i} \approx \frac{G^{-1}(1-\alpha) - G^{-1}(\alpha^c)}{G^{-1}(\alpha^c) - G^{-1}(\alpha)} = c^{-1+\hat{\xi}} \text{ holds.} \end{split}$$

The simple statistics are defined as

mean:

$$\left\{ \begin{array}{ll} \mu + \frac{\sigma(g_1 - 1)}{\xi} & if\xi \neq 0, \xi < 1\\ \mu + \sigma\gamma & if\xi = 0\\ \infty & if\xi \geq 1 \end{array} \right.$$

variance:

$$\begin{cases} \sigma^2 \frac{g_2 - g_1^2}{\xi^2} & if\xi \neq 0, \xi < \frac{1}{2} \\ \sigma^2 \frac{\pi^2}{6} & if\xi = 0 \\ \infty & if\xi \ge \frac{1}{3} \end{cases}$$

skewness:

$$\begin{array}{ll} \frac{g_3 - 3g_1g_2 - g_1^3}{(g_2 - g_1^2)^{3/2}} & if0 < \xi < \frac{1}{3} \\ -\frac{g_3 - 3g_1g_2 - g_1^3}{(g_2 - g_1^2)^{3/2}} & if\xi < 0 \\ \frac{12\sqrt{6}\zeta 3}{\pi^3} & if\xi = 0 \\ \infty & if\xi \ge \frac{1}{3} \end{array}$$

and kurtosis:

$$\begin{cases} \frac{g_{4-4g_1g_3-3g_2^2-12g_2g_2g_1^2-6g_1^4}{\left(g_2-g_1^2\right)^2} & if\xi \neq 0, \xi < \frac{1}{4} \\ \frac{12}{5} & if\xi = 0 \\ \infty & if\xi \geq \frac{1}{4} \end{cases}$$

with $g_p = \Gamma(1-p\xi)$ for $p \in [1, ..., 4]$, Euler's constant γ (Sondow (1998)), the Riemann zeta function $\zeta(\cdot)$ (Borwein et al. (2000)), and the gamma probability density function $\Gamma(\cdot)$.

Appendix C - Additional Figures and Tables

Table 3: UK Banking Sector (Largest Listed Banks): Individual Expected Losses (Put Option Values with and without Jump Diffusion),* 2005-2017 (In GBP Billions)

Table 3. UK Banking Sector (Largest Listed Banks): Individual Expected Losses (Put Option Values with and without Jump Diffusion), 2005-2017 (In GBP Billions)							
	Pre-Financial Crisis: June 2005– Aug. 2008	Global Financial Crisis: Aug. 2008– March 2010	Sovereign Debt Crisis: April 2010– Dec. 2011	Post-Crisis: Jan. 2012– Dec. 2017	Entire Time Period		
		Option Price	cing Model (with jump diffusion)				
			Bank 1				
25th percentile	0.48	1.90	0.55	0.17	0.36		
Median	7.77	35.27	1.15	0.42	0.86		
75 th percentile	28.96	57.94	13.36	0.86	3.08		
Average	18.46	33.60	6.05	0.70	10.13		
			Bank 2				
25 th percentile	0.01	0.20	0.11	0.01	0.02		
Median	0.21	15.51	0.13	0.04	0.11		
75 th percentile	1.26	22.10	0.27	0.12	0.27		
Average	2.61	13.19	0.18	0.08	2.40		
			Bank 3				
25th percentile	0.00	3.62	0.18	0.00	0.01		
Median	0.03	25.13	0.27	0.03	0.09		
75 th percentile	0.21	109.14	6.59	0.15	0.70		
Average	1.87	46.31	3.70	0.14	6.91		
			Bank 4				
25 th percentile	0.02	13.37	0.63	0.16	0.17		
Median	21.89	51.61	0.72	0.27	0.56		
75 th percentile	47.64	82.83	7.53	0.49	14.76		
Average	26.87	46.10	4.99	0.42	13.58		
	Option Pricing Model (without jump diffusion)						
			Bank 1				
25 th percentile	3.25	0.01	0.00	0.00	0.00		
Median	7.62	0.03	0.12	0.00	0.00		
75 th percentile	14.20	0.14	9.13	0.00	0.51		
Average	14.96	0.13	4.10	0.01	4.41		
0.5% VI		0.00	Bank 2				
25 th percentile	0.00	0.00	0.00	0.00	0.00		
Median	0.00	0.00	0.00	0.00	0.00		
75 th percentile	7.95	0.00	0.00	0.00	0.00		
Average	4.47	0.00	U.UU Dearly 0	0.00	1.14		
05th perceptile	0.00	0.00	Barik 3	0.00	0.00		
20" percenule	0.00	0.00	0.00	0.00	0.00		
75th percentile	0.33	0.12	0.00	0.00	0.00		
Лиогадо	10.20	0.41	4.19	0.00	U.IU 6 20		
average	22.39	0.27	2.07 Bank A	0.00	0.20		
25 th porcontilo	4.22	0.01	0.00	0.00	0.00		
20 percentile	4.22	0.01	0.00	0.00	0.00		
75th porcontilo	7.59	1.66	5.14	0.00	4.00		
Average	∠4.0U	1.00	0.14 0.61	0.01	4.09		
Average	17.99	1.50	3.01	0.01	5.29		

Source: authors' estimates. Note: Even though the sample starts on 1 January 2005, the first time window (June 2005-August 2008) above remains unchanged to facilitate the time-consistent comparison with the estimated joint expected losses, which are available from 21 June 2005 (based on a rolling window of 120 working days and daily updating).

Table 4: UK Banking Sector (Largest Listed Banks): Parameter Values of Fitted Multivariate Generalised Extreme Value (GEV) Distribution of Expected Losses, 2005-2017 (In GBP Billions)

	Pre-Financial Crisis	Global Financial Crisis	Sovereian Debt Crisis	Post-Crisis				
	June 2005– Aug. 2008	Aug. 2008– March 2010	April 2010– Dec. 2011	Jan. 2012– Dec. 2017	Entire Time Perio			
			All Banks					
			Average					
	6.60	29.01	1.46	0.55	5.82			
	1.85	14.92	1.20	0.36	2.70			
	0.17	-0.08	0.05	-0.06	0.01			
wness	0.25	-0.09	0.10	-0.09	0.02			
tosis	0.57	0.29	0.54	0.32	0.41			
)	0.47	0.57	0.48	0.56	0.53			
			Median					
	6.43	34.65	1.98	0.71	6.63			
	1.41	13.68	1.31	0.42	2.47			
	0.19	-0.07	0.08	-0.09	0.01			
ewness	0.24	-0.05	0.14	-0.14	0.01			
tosis	0.63	0.28	0.59	0.30	0.42			
))	0.43	0.48	0.44	0.47	0.45			
		Individual Banks						
			Bank 1					
	12.45	38.10	3.15	1.52	9.02			
	3.21	14.78	1.93	0.69	3.21			
	0.13	0.03	0.28	-0.15	0.00			
wness	0.59	0.35	0.61	0.27	0.41			
tosis	3.68	3.39	4.42	3.88	3.81			
			Bank 2					
	0.62	12.71	0.16	0.09	1.83			
	0.36	6.94	0.04	0.03	0.99			
	0.43	0.26	-0.42	-0.04	0.09			
wness	0.67	0.76	0.04	0.37	0.42			
tosis	5.08	3.43	2.95	3.69	4.03			
			Bank 3					
	0.35	41.38	1.58	0.63	5.78			
	0.23	21.63	1.15	0.47	3.13			
	0.36	-0.17	0.23	-0.07	0.08			
wness	0.66	0.10	0.67	0.36	0.44			
tosis	5.18	4.20	4.17	3.45	4.23			
			Bank 4					
	19.60	52.86	2.44	1.15	12.47			
	3.61	16.32	1.66	0.64	3.47			
	0.08	-0.49	0.31	-0.17	-0.07			
wness	0.34	-0.07	0.85	0.32	0.36			
tosis	3.35	2.72	4.58	3.95	3.80			

Source: authors' estimates. Note: The table shows the average and median parameter values of the multivariate generalised extreme value (GEV) distribution (mean (μ), scale (σ), and shape (ξ), which was fitted to the empirical distribution of individual expected losses with a non-parametric dependence structure (Y(ω)).

Table 5: UK Banking Sector (Largest Listed Banks): Joint Expected Losses under Adapted CoVaR Methodology, 2005-2017 (In GBP Billions)

Table 5. UK Banking Sector (Largest Listed Banks): Joint Expected Losses under Adapted CoVaR Methodology, 2005-2017 (In GBP Billions)							
	Pre-Financial Crisis: June 2005– Aug. 2008	Global Financial Crisis: Aug. 2008– March 2010	Sovereign Debt Crisis: April 2010– Dec. 2011	Post-Crisis: Jan. 2012– Dec. 2017	Entire Time Period		
		All	Banks, triggered by any bank				
		Condition	al Expected Shortfall (CoES) at 95% 1/				
25 th percentile	26.4	138.9	31.2	0.0	0.2		
Median	118.5	143.6	38.1	0.2	27.3		
75" percentile	195.5	1/3.3	117.7	23.2	115.1		
Average	131.0	134.2 Condit	b3.8	10.8	59.1		
25 th nercentile	25.1	32.9	31.0	0.0	0.1		
Median	115.6	140.8	37.5	0.2	26.1		
75 th percentile	123.7	166.8	111.9	22.4	109.2		
Average	121.8	121.1	60.1	9.9	54.8		
			All Banks, triggered by				
	Bank 1						
			CoES at 95%				
25 th percentile	25.9	140.0	64.7	0.0	0.1		
Median	117.5	146.5	118.8	0.1	35.1		
/ b" percentile	125.0	169.4	130.2	29.1	119.6		
Average	123.0	140.3	100.9 Col/aR at 05%	12.0	07.3		
25 th nercentile	23.7	136.0	43 0	0.0	0.1		
Median	114.7	141.8	118.5	0.1	26.0		
75th percentile	122.6	148.5	140.3	27.0	116.5		
Average	109.0	137.5	95.7	11.8	59.9		
	Bank 2						
			CoES at 95%				
25th percentile	23.2	142.2	31.9	0.0	0.4		
Median	109.7	146.5	111.6	0.2	31.1		
75 th percentile	262.2	183.0	132.6	23.0	127.5		
Average	141.6	159.7	89.6	12.2	77.2		
25th porcontilo	20.7	120.9	21 5	0.0	0.2		
20 ^{er} percentile Median	20.7	139.0	31.3	0.0	26.0		
75 th percentile	255.4	177.8	135.2	21.0	122.5		
Average	124.2	157.7	80.8	10.3	71.0		
	Bank 3						
			CoES at 95%				
25th percentile	69.6	140.3	30.0	0.1	14.6		
Median	195.5	142.9	33.4	0.2	31.4		
75 th percentile	275.7	165.6	39.6	21.6	107.6		
Average	169.6	156.9	40.9	11.6	71.8		
osti u		107.1	CoVaR at 95%				
25 th percentile	21.7	137.4	29.9	0.1	0.2		
ZE th perceptile	114.2	143.5	33.4 20 E	0.2	31.0		
Average	209.0	156.7	38.4	20.7	43.0		
Avolugo	Bank 4						
			CoES at 95%				
25th percentile	107.3	33.1	107.1	0.1	0.3		
Median	119.3	33.2	120.2	0.4	25.9		
75th percentile	254.6	33.3	143.4	22.0	119.1		
Average	140.6	33.2	110.2	10.4	70.5		
			CoVaR at 95%				
25th percentile	23.7	31.2	44.5	0.0	0.2		
Median	116.5	31.2	120.7	0.3	24.8		
75 ^{er} percentile	123.7	32.5	143.5	22.2	116.6		
Average	122.9	31.7	104.1	9.7	62.4		

Source: authors' estimates. Note: Conditional Value-at-Risk (CoVaR) measures the system-wide losses (as VaR at 95 percent) conditional on an individual stress event (i.e., a bank experiencing losses above a level consistent with a VaR of 95 percent, estimated over a rolling 120-day window). In applying CoVaR to CCA-generated expected losses, and for a complete comparison with the SES methodology, the individual stress level was also defined as CTE at 95 percent, estimated over a rolling 120-day window). In applying CoVaR to CCA-generated expected losses, and for a complete comparison with the SES methodology, the individual stress level was also defined as CTE at 95 percent, the changes in the total expected losses of all four sample banks indicate the systemic stress level (consistent with the definition of "the market" in the CoVaR methodology. The CTE at the statistical confidence level *a=95* percent was approximated via VA at 98.0 percent, the set of the systemic stress level (consistent with the definition of "the market" in the CoVAR methodology. The CTE at the statistical confidence level *a=95* percent was approximated via VA at 98.0 percent (based on the simplifying assumption of normally distributed expected losses, where the z-scores of (1-o)/(1-o) (for CTE) and Φ⁺ (o) (for VAR) are the same, where Φ and Φ⁺ are the standard normal probability distribution and quantile function).

Table 6: UK Banking Sector (Largest Listed Banks): Joint Expected Losses under Adapted SES Methodology, 2005-2017 (In GBP Billions)

Table 6. UK Banking Sector (Largest Listed Banks): Joint Expected Losses under Adapted SES Methodology, 2005-2017 (In GBP Billions)							
	Pre-Financial Crisis: June 2005– Aug. 2008	Global Financial Crisis: Aug. 2008– March 2010	Sovereign Debt Crisis: April 2010– Dec. 2011	Post-Crisis: Jan. 2012– Dec. 2017	Entire Time Period		
			All Banks (Sum)				
		Systemi	ic Expected Shortfall (SES) at 95% 1/				
25 th percentile	106.1	141.7	31.5	0.0	0.3		
Median	119.4	148.6	39.4	0.3	30.4		
75" percentile	254.3	191.0	119.0	24.7	116.6		
Average	139.0	142.4 Suete	b/. / amic Value_at_Rick (SVaR) at 05%	11.5	62.7		
25 th perceptile	27.0	33.8	32.4	0.0	0.2		
Median	118.3	163.0	38.9	0.2	30.9		
75 th percentile	253.9	188.7	119.3	25.0	115.3		
Average	134.7	133.7	67.1	11.3	61.1		
			Individual Banks				
	Bank 1						
		Margin	nal Expected Shortfall (MES) at 95%				
25 ^m percentile	40.1	36.5	0.0	0.0	0.0		
Xedian Z5th perceptile	46.8	44.4	0.0	0.0	0.3		
Average	49.8	43.3	27.4	0.3 7.0	41.8 20.9		
Average	40.0	40.0 Maro	inal Value-at-Risk (MVaR) at 95%	7.0	20.5		
25th percentile	26.9	0.2	0.0	0.0	0.0		
Median	46.1	53.5	0.0	0.0	0.2		
75 th percentile	73.7	63.9	27.6	0.3	41.1		
Average	48.6	39.7	10.8	7.1	20.3		
	Bank 2						
		Margin	nal Expected Shortfall (MES) at 95%				
25 th percentile	0.0	12.2	0.0	0.0	0.0		
Median ZEth perceptile	0.1	18.9	0.0	0.0	0.0		
75" percentile	70.0	21.7	0.3	0.0	0.1		
Average	21.3	70.0 Maro	inal Value-at-Risk (MVaR) at 95%	0.0	0.7		
25 th percentile	0.0	0.0	0.0	0.0	0.0		
Median	0.1	19.9	0.0	0.0	0.0		
75 th percentile	69.5	27.2	0.3	0.0	0.1		
Average	20.8	16.9	0.2	0.0	6.3		
	Bank 3						
		Margin	nal Expected Shortfall (MES) at 95%				
25 th percentile	0.0	78.5	29.9	0.0	0.0		
Median	0.0	88.9	33.3	0.0	0.0		
75" percenule	23.4	91.0	39.2	0.0	20.5		
Average	7 70 33.0 0.0 11.6 Marninal Value.at.Rick (MVaR) at 95%						
25 th percentile	0.0	0.3	30.4	0.0	0.0		
Median	0.0	87.3	33.8	0.0	0.0		
75 th percentile	23.1	91.2	39.8	0.0	26.6		
Average	7.1	63.3	33.9	0.0	11.9		
	Bank 4						
		Margin	nal Expected Shortfall (MES) at 95%				
25 th percentile	66.0	2.3	0.0	0.0	0.0		
Median 75 th perceptile	72.5	6.6	0.0	0.0	0.0		
/ 5" percentile	86.1	12.4	61.0	0.6	61.9		
Avelaye	59.8	11.8 More	∠3.U sinal Value_at_Risk (MVaR) at 0.5%	4.3	23.0		
25 th percentile	0.0	5.2	0.0	0.0	0.0		
Median	71.3	6.6	0.0	0.0	0.0		
75 th percentile	85.4	32.3	61.0	0.4	60.6		
Average	58.2	13.8	22.2	4.2	22.6		

Source: authors' estimates. Note: 1/ The Systemic Expected Shortfall (SES) aggregates (as weighted sum) the individual losses of all institutions ("marginal expected shortfall" or MES) conditional on a systemic stress event (i.e., the banking sector experiencing losses above a level consistent with its conditional tail expectation (CTE) of 95 percent, estimated over a rolling 120-day window). In applying SES to CCA-generated expected losses, and for a complete comparison with the CoVaR methodology, also a systemic stress level of VaR at 95 percent, estimated over a rolling 120-day window). In applying SES to CCA-generated expected losses, and for a complete comparison with the total expected losses of all incitate the systemic stress level of VaR at 95 percent was chosen as a threshold (which could be termed "Systemic VaR (SVaR)"). Given the concentration of the banking sector, the changes in the total expected losses of all to usample banks indicate the systemic stress level of usa a threshold (which could be termed via VaR at 95 percent). The CTE at confidence level a=95 percent was approximated via VaR at 95 opercent (base) of normally distributed expected losses of all to CPU of the changes in the total expected losses of all to CPU of the CTE at confidence level a=95 percent was approximated via VaR at 95 opercent (base).

Appendix D - Estimating the Market Value and Volatility of Equity of Unlisted Subsidiaries of Listed Parent Banks

The equity price, E, and equity volatity, σ_E , are essential inputs for the estimation of the marketimplied expected losses according to the CCA framework described in equation (26). However, in many countries, equity prices might not be readily observable or usable for significant parts of the financial system, such as unlisted savings and cooperative banks and a significant host banking sector, especially through locally incorporated subsidiaries and branches.

To resolve this issue, we specify a two-stage estimation approach based on the fundamental determinants of the price-to-book (PB) ratio of listed banks (i.e., the market value of equity expressed as a multiple of their book value) to derive both the equity price and equity volatility of unlisted banks (and local subsidiaries and branches whose contribution to the parent banks' group-level market valuation is not material).²⁹

The PB ratio for $j\epsilon m$ banks at time t is determined by the banks' fundamentals (profitability, asset quality, and credit rating) and general equity market conditions, and is specified by the following panel regression

$$PB_{j,t} = \alpha + \beta_1 RoE_{j,t} + \beta_2 LLR_{j,t} + \beta_3 \triangle Equity Market_{j,t} + \dots$$
$$..\beta_4 Dummy_GFC_{j,t} + \beta_5 Dummy_AA_{j,t} + \beta_6 Dummy_A_{j,t} + \beta_7 Dummy_BBB_{j,t} + \varepsilon_t$$
(60)

where LLR is the loan-loss reserve, proxied by the ratio of the share of reserves for impaired loans to total loans, $\triangle EquityMarket$ is the percentage change of the relevant (main) stock market index over the selected time horizon consistent with the frequency of available bank data, $Dummy_GFC$ is the dummy variable defined as one from mid-2008 (and zero otherwise), $Dummy_AA$, $Dummy_AA$, and $Dummy_BBB$ take the value of one for banks that are (1) rated "AA-" or higher, (2) rated between "A+" and "A-", and (3) rated between "BBB+" and "BBB-", respectively.

The PB ratio, in turn, is one of several explanatory variables of equity volatility

$$\sigma_{E_{j,t}} = \alpha + \beta_1 P B_{j,t} + \beta_2 \sigma_{EquityMarket_{j,t}} + \beta_3 \ln MarketCap_{j,t} + \beta_4 Leverage_{j,t} + \varepsilon_t, \tag{61}$$

where $\sigma_{E_{j,t}}$ is the exponentially weighted average annualised volatility, $\sigma_{EquityMarket}$ is the volatility of the main stock market index, $\ln MarketCap$ is the natural logarithm of the bank's market capitalisation, and *Leverage* is the book leverage ratio (i.e., the ratio between total liabilities and the book value of assets) (Campbell et al. (2001)).³⁰

We first obtain the time series of the return on equity, loan-loss reserve, and leverage ratio for all sample banks and relevant banking groups (from their financial disclosure as required by their

 $^{^{29}}$ This approach was first adopted when the Systemic CCA framework was applied as part of the bank stress test of the IMF's Financial Sector Assessment Programme for Hong Kong SAR (International Monetary Fund (2014)).

³⁰If the locally incorporated bank is not separately listed but accounts for a significant portion of total assets (i.e., 10 percent or above but less than 100 percent as "materiality" threshold), the market-based information of the banking group could reflect the operation of the local bank; however, some adjustments may be needed when using market-based information of the banking group. Thus, the potential equity value could be derived from multiplying the book value of the unlisted local bank by the actual PB ratio of the listed banking group; the equity volatility of the banking group would similarly apply to the unlisted local bank.

prudential regulator) and their credit ratings as well as the PB ratio of all listed banks/banking groups and the book value of equity of all unlisted banks.³¹ We supplement the banking information with the required market data, i.e., the annual percentage change of the main stock market index and its volatility. We can then estimate the coefficient values of the explanatory variables in equation (60) for all listed banks/banking groups can obtain $\hat{PB}_{j,t}$ of each unlisted bank by applying the coefficient values to the observed profitability, asset quality, and credit rating. Multiplying the book value of equity of the local bank by $\hat{PB}_{j,t}$ yields the implied equity value, which is then used to derive the quasi-market capitalisation $\ln MarketCap_{j,t}$ as input to equation (61) in order to derive $\hat{\sigma}_{E_{j,t}}$ (with $\hat{PB}_{j,t}$ and the leverage ratio as additional bank-specific inputs).

This approach is applied to two different cases: (i) neither the local bank nor the banking group (if applicable) are listed, or (ii) the local bank is not listed and accounts only for a small portion (i.e., less than 10%) of the total assets of the listed banking group, whose equity prices (and more generally market-based information) has limited relevance for the market-implied default risk of the local bank.

 $^{^{31}}$ Jordan et al. (2011) examined the PB ratio for 6,604 bank stock observations between December 2006 through June 2009. They relate each bank's PB ratio to several fundamental ratios and find that lower relative costs, higher non-interest income, and lower assets in non-accrual or foreclosed status are associated with higher PB ratios (while controlling for size and other attributes).